FACTORS AFFECTING STUDENTS' ACADEMIC PERFORMANCE IN VISUAL ARTS EDUCATION IN GHANA: AN EVIDENCE FROM DORMAA SENIOR HIGH SCHOOL.

by

Anna Yeboaa

(BEd. Art)

A thesis submitted to the Department of Educational Innovations in Science and Technology, Kwame Nkrumah University of Science and Technology in partial fulfilment of the requirements for the degree of

MASTER OF PHILOSOPHY IN ART EDUCATION

Faculty of Art

College of Art and Built Environment

February, 2018

© 2018, Department of Educational Innovations in Science and Technology

DECLARATION

I hereby declare that this research is my work towards the degree of MPhil. Art Education. It does not contain previously published academic resources by any individual nor approved for the award of Degree at any other University. Every source referred to in this study has been appropriately cited and acknowledged.

KNI	JST
Anna Yeboaa (PG3640215)	
Student's Name and Number	
(Signature)	(Date)
Dr. P. Osei-Poku	
(Supervisor)	(Date)
Dr. P. Osei-Poku	
(Head of Department)	(Date)

ABSTRACT

This study was conducted to examine factors affecting students' academic performance in Visual Arts Education in Ghana with an evidence from Dormaa Senior High School in the Brong Ahafo Region. The study's objectives were: to identify factors that result in high students' academic performance of Visual Arts education in Dormaa Senior High School; to examine specific factors that underpin low academic performance of Visual Arts students in Dormaa Senior High School; and to propose strategies that can be put in place to enhance academic performance of Visual Arts students in Dormaa Senior High School. As a descriptive research under qualitative study, a sample size of forty (40) respondents were purposively and conveniently selected to help in eliciting responses for this study. The methods of data collection were focus group discussions and unstructured interview. The sources of data were primary and secondary. Data were analysed using a thematic approach. The study's findings revealed that there are factors that affect academic performance of Visual Arts Students in Dormaa Senior High School. The study further revealed that for the past three years (2014-2016) Dormaa Senior High School has recorded low academic performance for its Visual Arts Students. Finally, useful insights gained from the qualitative analysis as well as the findings and conclusions could serve as practical evidence for school administrators interested in adopting strategies in enhancing academic performance of their students.

ACKNOWLEDGEMENTS

I wish to acknowledge the people who greatly contributed to the accomplishment of this research project. I greatly appreciate my family for their moral support, patience and understanding.

I thank my classmates for their moral support and encouragement. Sincere thanks to my supervisor, Dr P. Osei-Poku, for his enabling guidance, patience, constructive criticism and personal interest in the progress of my study, his availability in reading through the proposal and preparation towards my project. I will always remember his humbling remarks and academic input towards my project work.

I am also greatful to all the lecturers at Art Education section who gave advice and suggestions at various stages of my research. Special thanks go to the students, staff of the visual arts department and the entire staff of Dormaa Senior High School for their support and co-operation during the period of study.

DEDICATION

I dedicate this research to my mum, Madam Margaret Abenaa Fosuaa, for her love and support throughout my education.



TABLE OF CONTENTS

DECLARATION	ii
ABSTRACT	iii
ACKNOWLEDGEMENTS	iv
DEDICATION	V
TABLE OF CONTENTS	vi
LIST OF TABLES	x
CHAPTER ONE	1
INTRODUCTION	1
1.1 Overview	1
1.2 Background to the Study	1
1.3 Statement of the Problem	3
1.4 Research Objectives	6
1.5 Research Questions	7
1.6 Limitations of the Study	7
1.7 Delimitation	9
1.8 Definition of Terms	9
1.9 Significance of the Study	10
1.10 Organization of the Rest of the Text	
CHAPTER TWO	
REVIEW OF RELATED LITERATURE	12
2.1 Overview	12
2.2 Theoretical Framework	12
2.2.1 Critical Theory	14

	2.2.2. Systems Theory Input-output Model	. 15
	2. 3 Overview of Academic Performance	. 15
	2.4. Visual Arts Education Programme in Ghana	. 18
	2.5. Factors that Affect Academic Performance of Students	. 20
	2.5.1 School's Leadership	. 21
	2.5.2 Teachers' Characteristics	. 23
	2.5.3 Students' Characteristics that Affect Academic Performance	. 25
	2.5.4. Type or Nature of School	. 25
	2.5.5. Students' Learning Preferences	. 26
	2.5.6. Class Attendance and Academic Performance	
	2.5.7 Gender Factors	. 28
	2.5.8 Entry Qualifications and Prerequisites	. 29
	2.6 Parents or Guardians Factors	. 29
	2.7. Home Conditions of Students that Affect Academic Performance	. 30
	2.8. Strategies that can be put in Place to Improve Students' Academic Performance	
СН	APTER THREE	. 35
1	METHODOLOGY	25
ľ	METHODOLOG1	. 33
	3.1 Overview	. 35
	3.2 Research Design	. 35
	3.3 Research Approach	. 36
	3.3.1 Qualitative Approach	. 36
	3.3.2 Qualifications of Qualitative Approach	. 36
	3.4. Research Methods	. 38
	3.4.1 Case Study	. 38
	3.4.2 Descriptive Method	. 41
	3.5 Population and Sampling	. 42
	3.5.1 Population	. 42
	3.5.2 Sample Selection	. 43
	3.5.3 Sample Size	12

	3.5.4 Sampling Method: Non-probability Sampling	. 45
	3.6 Sources of Data	. 46
	3.7 Data Collection Methods	. 47
3.	7.1 Focus Group Discussions	. 47
		40
	3.7.2 Unstructured Interview Method	
	3.7.3 Pilot Study	
	3.8 Data Analysis	. 51
	3.9.1 Reliability	
	3.9.2 Validity	. 54
	3.9.3 Reliability and validity of the Study's Results	
CHA	APTER FOUR	. 56
ΡI	RESENTATION AND DISCUSSION OF FINDINGS	. 56
	4.1 Overview	. 56
	4.2 Objective 1: Factors that Result in High Academic Performance of Visual Ar Students in Dormaa Senior High School	
	4. 2.1: School based Factors / School Leadership	. 57
	4.2.2 Teachers' Factors	. 59
	4.2.3: Student's Characteristics	. 61
	4.2. 4: Parent's/ Guardian Socio-economic Conditions	. 62
	4.2. 5: Home Conditions or Parental Support	. 63
	4.3. Objective 2: This comprises categories of responses that underpin factors that account for low academic performance of Visual Arts Students in the school	
	4.3. 1: School Environment	
	4.3.2: Teacher Factors	
	4.3.3 Student's Characteristics	
	4.3.4: Parent's/ Guardian Socio-economic Condition	
	4.3.5: Home Conditions or Parental Support	
	4.4. Objective 3: Strategies to be adopted to enhance effective academic	. , 0
	performance of Visual Arts Students in Dormaa Senior High School	. 71

	4.4. 1: The Role to be Played by the Leadership of Dormaa Senior High School in Adopting Strategies to Enhance Academic Performance of Visual Arts Students in the Visual Arts department.	n
	4.4. 2: Specific Roles to be Played by Teachers in Adopting Strategies to Enhance Academic Performance of Visual Arts Students	
	4.4.3: Specific Roles to be Played by Visual Arts Students' themselves in Adoptir Strategies to Enhance their Academic Performance.	_
	4.4. 4: Specific Roles to be Played by Parents and Guardians in Adopting Strateg to Enhance the Academic Performance of Visual Arts Students	
	4.4. 5: Specific Roles to be Played by the Dormaa Ahenkro Community in Adopting Strategies to Enhance the Academic Performance of Visual Arts Studen	
	4.4. 6: Specific Roles to be Played by Old Students in Adopting Strategies to Enhance the Academic Performance of Visual Arts Students	87
CH	APTER FIVE	90
	UMMARY, CONCLUSION AND RECOMMENDATIONS	
	5.1. Overview	
	5.2.1 Summary of Findings	90
	5.3 Parents and guardians must adopt the Following strategies to enhance the academic performance of Visual Arts Students:	94
	5.4 The Dormaa Ahenkro Community should adopt the following strategies to enhance the academic performance of Visual Arts Students	95
	5.5 Old students need to adopt the following strategies to enhance the academic performance of Visual Arts Students	96
	5.6 Conclusions	
	5.7 Recommendations for Policy Formulation	99
	5.8 Suggestions for Further Studies	02
REF	FERENCES 1	04
APF	PENDIX 1	18
ΛDE	PENDIX 2	21

LIST OF TABLES

Table 3.1: Number of respondents	4
----------------------------------	---



CHAPTER ONE

INTRODUCTION

1.1 Overview

This chapter provides an introduction to the thesis. It comprises the following subtopics: Background to the Study, Statement of the Problem, Objectives and Research Questions. It spells out the Limitation, Delimitation, Definition of Terms, the Significance of the Study and the Organization of the Rest of the Text.

1.2 Background to the Study

Education is described as a globally recognized basic human right, which can be seen as a form of investment that contributes to the development of both individuals and the society. Specifically, formal education is said to contribute highly to social-economic development and social mobilization in any society (World Bank Group, 2009).

The basis for any development is said to begin with the development of human resources. Partaking in examinations plays important contribution in area of developing human resources for any given nation(World Bank, 2009; and UNESCO, 2007). The central objective of every education is to get individuals prepared for the job market in the form of transferring virtues such as knowledge, skills, attitudes and cultural norms of the adult world for the younger generations (Griffin, 1998).

Also, it is one of the most powerful instruments known for reducing poverty and inequality and for laying the basis for sustained economic growth; raises the productivity and efficiency of personnel. Education provides solid processes through

which key indicators such as intellects, moral capacities, proper conduct, and technical competencies inherent in learners are developed to make them fit in the larger society (Tuan, 2009, as cited in Dorleku, 2013). There is positive correlation between teaching and learning since the two form core part of education, provide essential learning tools and the basic learning content required by human beings to survive, to develop their full capacities, live and work in dignity, participate fully in development, improve the quality of their lives, to make informed decisions and to continue learning (United Nations Educational, Scientific and Cultural Organization [UNESCO], 2000 as cited in Siaw, 2009).

Education empowers people, opens up countless number of opportunities, reduces and prevents the outbreak of diseases and eradicates poverty and helps individuals to acquire a greater voice in society, whereas on the perspective of nations, education opens doors to economic and social prosperity which is always spurred by a dynamic workforce and well-informed citizenry who can compete and cooperate willingly on the global scale (World Bank, 2000).

Across the globe, investments and contributions done in education can be divided into three levels: primary, secondary and tertiary. In specifics, secondary education is mainly aimed at meeting global challenges in areas related to science and technology as well as organizations of production processes and markets (Mbelle and Katabalo, 2003).

It is stressed that investments done in education can result in faster growth for both developed and newly industrialized countries (Schultz, 2002).

Little wonder developing countries, especially those found in the Sub-Saharan Africa have started investing in education from the levels of primary, secondary and tertiary in the form of increasing enrolment and improving education quality.

A study conducted by Miller-Grandvaux and Yoder (2002) on secondary school education revealed that Secondary schools are key – in terms of providing interventions in education in Sub-Saharan Africa. However, it can be maintained that the main challenges bedeviling secondary school education are said to be academic performance of students. Cambridge University Reporter (2003) stresses that academic performance is frequently defined in terms of examination performance. In this current study, academic performance was considered by senior high students' performance in tests, in course works and performance in West African Senior School Certificate Examination (WASSCE).

Students are dynamic people always in progress. They are full of potentials, which have to be discovered and properly channeled hence the inclusion of Art Teaching Curriculum in our school systems.

1.3 Statement of the Problem

Education is one of the most important aspects of human resource development. Failure in the national examinations, especially in the West African Senior School Certificate Examination (WASSCE) spells doom for the students whose lives become uncertain and full of despair. Therefore, a student's life is determined by academic performance in the national examinations. Public pressure on school administrators and teachers to improve

academic performance has led to schools coming up with various performance improvement strategies, including extra supplementary tuition, reward and punishment systems for well performing and poor performing students, forced grade repetition among others. However, some of the strategies employed by schools to improve academic performance are not grounded in research evidence, while some like grade repetition and extra supplementary tuition have been shown to be counterproductive (Bray, 2007).

Besides, academic performance of Senior High School students is measured through tests, class exercises and other ways for continuous assessment purposes in the elective Business, General Arts, Science and Visual Arts subjects and in Core English, Integrated Science, Core Mathematics and Social Studies, which constitute the core curriculum. High academic performance in the core and elective curriculum subjects is required of Senior High School students over the three-year period of Senior High School education.

A student must make a cumulative score of WASSCE Aggregate 24 or better in the core and elective subjects to be considered for admission to higher education programmes (Curriculum Research and Development Division of Ghana Education Service, 2008; Asihene, 2009). A number of international studies on academic performance, including Ehren and Visscher (2008), Chaturvedi (2009), Lloyd and Ghuman (2010) have been carried out and have shown that various factors influence academic performance. Also, a number of local studies such as Daniels (2005), Kitavi (2005), Katana (2007), Orenge (2007) among others have been carried out. Several factors have generally been identified as causes of poor academic performance. Orenge (2007) found a lack of in-

service training, inadequate teacher supervision by head teacher, inadequate staff meetings, low level of parents' education as causing poor academic performance. Katana (2007) found that inadequate teaching staff and high pupil enrolment, poor syllabus coverage, poor time management, poor standards of discipline in the school, poor teacher, pupil and parent relationship, lack of motivation for both teachers and pupils and inadequate teaching and learning materials cause poor academic performance.

Failure in the national examinations, especially in the West African Senior School Certificate Examination (WASSCE) spells doom for the students whose lives become uncertain and full of despair. West African Senior School Certificate Examination performance determines whether the students will proceed to university or to other tertiary institutions. Therefore, a student's life is determined by academic performance in the national examinations. The relevance of this is that, secondary school administrators, including that of Dormaa Senior High School, are pressurized to improve upon the grades obtained by students in WASSCE. Public pressure on school administrators and teachers to improve academic performance has led to schools coming up with various performance improvement strategies, including extra supplementary tuition, reward and punishment systems for well performing and poor performing students, forced grade repetition among others. It is in line with this that this research also expresses concern about the kind of grades that the students are admitted into the school with, thus leading to poor academic permormance in WASSCE.

The poor academic performance of Visual Arts students of Dormaa Senior High School (DSHS) has been a concern for quite some time. Visual Art was first introduced in Gold

Coast now Ghana in the year 1908 (Foster, 1967). The Elective Visual Arts Courses in the case of Dormaa Senior High School are General Knowledge in Art, Graphic Design and Leatherwork. The 2014 study of WASSCE results gives the following results picture. The school presented 35 visual arts students for WASSCE. Out of this number, 28 students (80 %) were successful in all the electives whereas seven (7) students failed in the elective. The 2015 WASSCE result also gives the following appalling results picture. The school presented 38 students for Visual Arts students. Out of this number, ten students (24 %) were successful in all the electives subjects whereas twenty-eight failed the elective. The 2016 WASSCE equally indicated that out of 29 students presented for the examination, Six students (21 %) passed all the three Elective Visual Arts Courses. This indicated that the Dormaa Senior High School visual arts students result is deteriorating yearly. The present study therefore sought to identify factors that affect academic performance of visual Arts students' of Dormaa Senior High School in Dormaa Central Municipality of the Brong Ahafo Region of Ghana. The students who pass and are able to progress to the tertiary level become assets to the nation while those who fail in the final examination become burden to their families, their society and the nation at large.

1.4 Research Objectives

- 1. To identify factors that result in high students` academic performance of visual arts education in Dormaa Senior High School.
- To examine specific factors that underpin low academic performance of Visual Art Students in Dormaa Senior High School.

3. To propose strategies that will enhance academic performance of Visual Art students in Dormaa Senior High School.

1.5 Research Questions

- 1. What are the factors that result in high students' academic performance of visual Art students in Dormaa Senior High School?.
- 2. What are the specific factors that underpin low academic performance of Visual Arts students in Dormaa Senior High School?
- 3. What are the strategies that will enhance academic performance of Visual Arts students in Dormaa Senior High School?

1.6 Limitations of the Study

It is instructive to note that all academic studies like this study cannot do without limitations that always confront the researcher. Specifically, the study was prone to the following limitations.

Responses that were obtained from the respondents were validated through authoritative documented sources such as attendance register, terminal reports, placement forms for admissions, WASSCE detailed results for the school. Again, there was a clear unwillingness on the part of certain respondents for both the focus group discussions and the interview to make themselves available to provide right responses or information to help further the study. This reduced the actual number of respondents with the right information to provide to the researcher to help in successfully carrying

out the study. Others who made the attempt to respond to the interview question did not respond to certain questions. This also affected the overall quality of the responses the researcher was expecting.

The adoption of interview as a data collection device led to the following as limitations. Participants feared to committing themselves to telling the truth. The researcher had no control over responses considered as bias. It was difficult to trust what the respondents actually said during the interviews. The answers could be influenced by their colleagues who were not ready to respond to certain questions. Related to this is that other respondents booked for interviews declined audience when the time was due.

Interviews are argued as highly time consuming activities and filling of data overload as a result of the huge volume of rich data produced (Karen, 2011). Moreover, Easterby (2002), argued that understanding issues from an interviewee's point of view can be extremely difficult, especially when the respondent himself is unable to articulate views or unwilling to divulge sensitive information.

Lastly, the subject, factors affecting academic performance is broad. It was not possible for the researcher to explore all the variables that underpin it, hence the three objectives that were thoroughly researched into. Also, this study covers only one Senior High School, it would be difficult to generalize the findings to the nation as a whole.

Upon these limitations, the quality of responses from the respondents and the evidence of data gathered from the students and teachers in Visual Arts department, and the inability of some respondents to divulge some information, it did not affect the quality of the work.

1.7 Delimitation

Geographically, this study was limited to Dormaa Senior High School. Study participants were students and teachers of the Visual Arts Department. Participating year's batch of students was selected from those that have been consistently performing well for the period 2014 - 2016 and those that have been performing poorly for the same period.

Contextually, this study focused on the factors that affect students' academic performance of visual arts education. The primary data were gathered through the use of focus group discussion and interview with respondents purposively selected.

1.8 Definition of Terms

- Hermeneutics: The study or theory of the methodical interpretation of text,
 especially holy text.
- Invigorated: feeling stronger, healthier, and more full of energy.
- Unearthing: discovering
- Tenets: an opinion, belief, or principle that is held as absolute truth by someone or especially an organization
- Deductive analysis: inferences made from general principles
- Dross: irrelevant material
- Efficiency: effectiveness
- Extant: still existing despite being very old: surviving

 Spurious: a statement that is unreasonable because it is not based on true facts or a sensible way of thinking.

1.9 Significance of the Study

The study is considered as significant and timely because the government, parents and other stakeholders in education spend large portions of their respective resources in education. Low levels of Visual Art performance lead to undesirable wastage and deny students' entry into their preferred programmes when furthering their professions (usually very competitive). The effect spills over to the labour market, where the students fail to get employed in lucrative jobs.

School administrators, who include Headmasters, assistant headmasters and heads of department, the study provides data that could be used to improve management practices for improved academic performance of Visual Arts students. By establishing the extent to which schools are implementing the seven correlates of effective schools and how this influences academic performance, the study could stimulate strategic thought among school administrators. By identifying the strategies employed by well performing schools, poor performing schools could learn lessons that could enable them improve academic performance of their students. Senior High school students could benefit from the study because findings reveal the way students' performance can be affected by the facilities and members of the school community and the school environment. Through this, students in poor performing schools could be encouraged to adopt strategies employed by well performing schools in order to improve academic performance. The study is also significant to the community that their investment in education is expected

to translate to quality education, and the findings show how this can be achieved. The study also adds to the existing body of knowledge on the determinants of academic performance.

1.10 Organization of the Rest of the Text

Chapter Two covers the review of literature relevant to the topic. Chapter Three deals with the strategies adopted in the research design, data collection, the sample and sampling technique, administration of research instruments and data analysis plan. Chapter Four presents the discussion and analysis of findings while Chapter Five presents the summary, conclusions drawn from the study and recommendations for improving the situation in the schools.

CHAPTER TWO

REVIEW OF RELATED LITERATURE

2.1 Overview

This chapter presents related literature on the theoretical framework on performance, Overview of Academic Performance; Visual Arts Education programme in Ghana; Factors that affect low academic performance of the students, Strategies for enhancing students academic performance.

2.2 Theoretical Framework

Academic performance is one of the most extensively used concepts in educational research and assessment in second cycle institutions. Bandura (1977) worked to develop the theory of Self-efficacy in order to elucidate, predict and affirm the fact that the theory has an effect on academic performance.

The Self-efficacy theory (Bandura, 1977 and 1986) functions as a theoretical framework of this research since it refers to a person's beliefs to effectively accomplish a course of action necessary to achieve a preferred outcome in order to elucidate academic performance (Greene, Miller, Crowson, Duke and Akey, 2004; Miller, Greene, Montalvo, Ravindran and Nichols, 1996).

Most recent study from Greene et al. (2004) demonstrate a substantial influence of self-efficacy on academic achievement. According to Bresó, Schaufeli, and Salanova (2011), self-efficacy affects academic duties and performance and the effective utilization of the knowledge and skills acquired. The model is estimated to have an indirect influence on

the academic performance through educational approaches and a direct influence on academic performance.

Bandura (1986: 395) defines the theory of "self-efficacy" as "people's judgment of their capabilities to organize and execute courses of action required to attain designated types of performances." According to Bandura (1986 and 1997); and Schunk (1991), these competencies is causal elements of educational motivation choice and performance.

Vrugt, Langereis, and Hoogstraten (1997); and Vrugt, Oort and Zeeberg (2002) reaffirm the belief that high self-efficacy will inspire a person to follow inspiring individual goals and expend considerable determination to realize them, and further demonstrate high academic performance, while low self-efficacy will bring about the pursuance of a less high level of academic achievement. In respect of this, learners who are confident that they "are able and that they can and will do well are much more likely" to be invigorated regarding determination, perseverance and conduct than those who are confident "they are less able and do not presume to be successful (Pintrich and Schunk, 2002). There is also evidence to suggest that these confident students will also be more cognitively engaged in learning and thinking than those who doubt their capabilities to do well (Pintrich, 1999; Pintrich and Schrauben, 1992).

One of the findings of Salomon's (1984) study establishes that self-efficacy is completely linked to self-rated cognitive effort in the course of students' learning from writing material, which was assumed to be difficult. Learners who are not self-assured or see themselves not capable may escape conducts that appear to be taxing (Bandura, 1993). Grounded on the above-mentioned realistic outcomes in relation to Bandura's

theory, it is probable that self-efficacy has a positive effect on understanding meaning (deep), and a negative effect on knowledge retention (surface). This is buttressed by Malka and Covington (2005) and Zimmerman and Kitsantas (2005) who claim that self-efficacy reliably emerge as a robust predictor of academic performance.

2.2.1 Critical Theory

The academic performance of students nowadays are used to determine the effectiveness of how quality a school is and the effectiveness of its teachers. They are sometimes isolated from other education stakeholders like policy makers, educators and the parents. Employers sometimes threaten teachers with job losses anytime students' academic performance in terms of results are not satisfactory to parents and the public at large.

In this thesis critical theory is used to comprehend the nature of problems attributed to the Visual Arts students' low performance in schools. Critical theory hinges on the premise that respondents are being humans. Critical theory is premised on the fact that humans should have the liberty they require and should be informed about the behaviour accepted in a particular society (Basit, 2010). This theory is also explained by Standford Encyclopedia (2015, p. 1) that critical theory "provides the descriptive and normative bases for social inquiry aimed at decreasing domination and increasing freedom in all their forms". It appears that critical theory empowers individuals who are less privileged. As a result, tutors are criticized due to low academic performance since the onus lies on them to ensure students go through all forms of testing.

"In critical theory, the viewpoint of the researcher is to find out about common meanings and get understanding into a context, for example, school, classroom."

(Wellington, 2000, p. 16). Educators in this theory usually learn about what is significant about a group of people and then perceiving it from their angle (Neuman, 1997). The theory seeks to examine the false nature of the arguments presented by an individual or a group making it weak and also by posing questions to that (Cohen, Manion and Morris, 2007). Still, this theory has been used by teachers who have limitations in carrying out activities which hitherto might help improve the academic performance of students who are unable to do so.

2.2.2. Systems Theory Input-output Model

System theory which made use of the input-output model was propounded by Ludwig Von Bertalanffy in the 1950s. This theory, as reported by Koontz and Weirich (1988) argues that an enterprise does not work in a vacuum, but is hinged on external conditions which imply that it receives inputs, and generates outputs for onward transmission for its surroundings. Applying this theory to the current study, the selected school admits (inputs) and then changes them through the process of teaching and learning which is expected to be reflected in the academic performance of students' (output).

2. 3 Overview of Academic Performance

In every educational institution, accomplishment or attainment of success in difficult subjects is measured and assessed by academic performance as a criteria. As noted by Cambridge University Reporter (2003), "academic performance is usually based on students' performance or success in their examination". Relating this definition to the

study, the academic performance of students takes the form of test, class work and examinations at the senior high school level.

Academic performance depicts the way in which students go about their studies, and how they manage their work as per their teachers, which is related to content and intelligence. It is also based on a student's competence (Osei-Mensah, 2012).

Ankomah (2002) is of the view that in every educational setting, the ultimate aim is to achieve academic excellence or how students cope with their academic work. It also involves how students can manage their work from teachers and the way teachers are able to arrive at their aims.

The aim of education in line with academic performance is students ability to achieve a goal when tested on what they have been taught (Otoo, 2007), which could be linked to the curriculum content of the Visual Arts programme, the learner's intellect, and its dependence on student's ability.

Babatunde and Olanrewaju (2014) argue that academic performance is the same as academic achievement. Aremu and Sokan (2003), as cited in Osei Mensah (2012) believe that academic performance of students is not only based on a pointer to the effectiveness or otherwise of schools, but it is somewhat a main factor likely to affect the future of youths at schools and in the society to a larger extent.

Adams (2000) as cited in Yeboah (2014) has emphasized the need for a number of ways to evaluate the academic performance of which examination is one of the factors when considering quality education. In order to measure academic performance, examination marks which indicate the strength of a particular student is crucial in determining the

knowledge and skills acquired by a student. With the issue on student grading, students exhibit their knowledge when they partake in a number of written and oral exercises, presentations, home works, as well as class exercises and discussion. Performance of students is ranked in grades or letters which correspond to a particular score, and this suggest the performance of the student, and helps to check the teacher's and school's performance as against the grade of the student (Bell, 2012). Moreover, evaluation of a student is based on standardized tests which should be in line with the objectives of the subjects being studied. Previously, academic performance was evaluated by observations from a teacher and this constituted a big percentage of a student's assessment, however, at present, a sum of how well a student performs seems to be the new approach to this issue (Ali, Haider, Munir, Khan1, and Ahmed, 2013). Considine and Zappala (2002) highlight the fact that students whose parents are within a high social class have the opportunity to get academic materials which help to improve their understanding. Graetz (1995) also supports the opinion that parents background can also enhance students academic performance.

The performance of a student is affected by factors such as their admission scores, social class of their parents or guardians, previous school attended (Kyoshaba, 2009), student's sex, age, residence of the student, mode of instruction, methods of teaching, hour of studies, the accomodation status of students (day or boarder) (Ali, Haider, Munir, Khan1, and Ahmed, 2013 According to Considine and Zappala (2002), students whose parents background are good in terms of social, educational and economical are likely to provide the necessary materials for their wards at school leading to a high academic performance.

They add, the social class of parents is based on their educational background, type of employment, and income level. For this study, the social class is based on income level of parents and their educational and employment status. In a similar vein, Graetz (1995) has observed that students from a high social class background perform better than those from a low social class bracket.

2.4. Visual Arts Education Programme in Ghana

In Ghana, Visual Arts as a programme is studied by senior high school (SHS) students and tertiary levels. At the SHS level, the vocational skills programmes comprise Visual Arts and Home Economics. The curriculum for students reading Visual Arts include Sculpture, Ceramics, Graphic Design, Picture Making, Textiles, Jewellery, Leatherwork, Basketry, and General Knowledge in Art which serves as core to all students (Evans-Solomon, 2004; Asihene, 2009). The Curriculum Research and Development Division's Teaching Syllabus for Visual Arts for General Knowledge in Art (2008) suggests that students should have adequate foundational knowledge and skills in any of the eight elective subjects in addition to self-employment or apprenticeship with respect to students who wish to finish their education at the second cycle.

The syllabus offers advice on two types of elective subjects to choos9 e from, for each student: namely, two-dimensional and three-dimensional subject. Subjects under the two dimensional category include Graphic Design, Picture Making and Textiles. Those classified under three dimensional art forms comprise Sculpture, Ceramics, Jewellery, Basketry and Leatherwork. In a related study, Siaw (2009) has reiterated that three elective subjects (one of the three-dimensional category and two from two-dimensional

category) plus a compulsory core subject: General Knowledge in Art, making it a total of four elective subjects. By studying these subjects for a period of three years in addition to other core subjects such as English Language, Mathematics, Integrated Science and Social Studies, students get the chance to write their final year exams, West African Senior Schools Certificate Examination (WASSCE) and hopefully get the opportunity to continue their education at the tertiary level or enter the job market.

The choice of a particular Visual Arts subject is dependent on specialist teachers available in a school and the right facilities available, tools, equipment and relevant materials (Owusu-Afriyie, 2009). It is worth noting that the range of subjects studied by students at the SHS level will to a large extent has an impact on a student's academic and career path after completing the programme.

General Knowledge in Art (GKA) as well as other subjects has its own challenges (Evans-Solomon and Opoku-Asare, 2011). The 2008 Teaching Syllabus describes GKA as a composite subject which has been taken from all the Visual Arts subjects studied at the Senior High School level and is supposed to give information on the history of art, creativity and appreciation, the elements and principles of art, and practical skills in various exercises. Just like any other elective subject, GKA involves theory and practice-based topics. The first part, which is the theoretical component is needed to advance the scope of art vocabulary so as to equip students with the right knowledge to communicate in the subject area with reference to art. The practice-based aspect of the syllabus is to underscore what has been learnt in the individual subjects (Evans-Solomon and Opoku-Asare, 2011).

Opoku-Asare (2008) asserts that GKA has no specialist teachers, hence, every teacher teaching any of the elective subjects is likely to teach the subject well. In order to teach GKA, it assumes that a teacher should be knowledgeable, resourceful, genius, helpful nature of teachers who are assigned to teach the subjects in various schools. Despite, Visual Arts could help transform secondary education by providing jobs for individuals (Rihani, 2006), it seems that Ghana might not be able to benefit from this programme if the vocational and technical is under resourced and thereby resulting in a negative impact on human capital development for economic growth (The President's Education Reform Review Committee, 2002).

2.5. Factors that Affect Academic Performance of Students

Many researchers have argued that there are number of factors affecting students' performance at various schools, colleges and even universities (e.g. Durden, and Ellis, 1995; Kwesiga, 2002; Crosnoe, Johnson, and Elder, 2004; Kyoshaba, 2009; Osei-Mensah, 2012; MolokoMphale, and Mhlauli, 2014). Findings from this research suggest students' effort, previous school attended, the background of the parent, income of the family, self-motivation of students, the age of the student, students learning choice and grade of student are factors that have an effect on students academic performance. The relevance of these studies by previous authors should lead to corrective measures that is likely to improve the academic performance of SHS graduates. So it stands to reason that students who perform well might be encouraged to pursue further studies after completing their first degrees and even move on to doctorates based on a person's career path.

2.5.1 School's Leadership

The success of an institution is dependent on the leadership of that institution. Currently, leaders of schools may perform multiple tasks as compared to their counterparts in the past. Their roles include managing, giving instructions, curriculum delivery, counselling and performing advisory roles in the society. Experts in the field think that there is a relationship between good leadership and school performance (Emerson and Goddard, 1993). Some authors, for instance, Yusuf (2012) acknowledges that no institution is greater than their leaders and vice versa. On the one hand, a strong relationship exists between the quality of school and the quality of its leadership. On the other hand, some authors seem to argue that there is not much evidence to substantiate that relationship between the two variables (Witziers, Bosker and Kruger, 2003). MolokoMphale and Mhlauli (2014) think that many studies about school leadership quality impacts a school's success, teacher's effectiveness and students' academic effectiveness. The head of a school is in-charge of all activities that occurs in a school and is accountable to Educational Directorate, Board of Governors, Parents- Teacher Association (P.T.A.), among others. It is the duty of a head to monitor the success of a school by making teaching and learning satisfying and more effective. It is the desire of a head of school that he or she is trusted, respected and praised by the community he or she works in. A school's reputation in terms of academic performance could be poor and if this happens, then the school's reputation is diminished. This is what most of the times, the head of a school tries to avoid. Academic performance is likely to be maintained by a guiding philosophy for a school, instructions for a school with its settings and aiming to achieve its target as provided by a supervisory body, that is, becoming instructional leaders.

Among the ways in which leaders in a school have an impact on students' performance include instructional leadership which consists of goals setting, curriculum management, lesson plans, monitoring, resource allocation and assessment of teachers on a regular basis to help students learning and development (Concordia University, 2013). Effective instructional leaders have the ability to build the art of learning, make provision for resources towards quality teaching, and structure the classroom environment to encourage teachers and students' motivation. Indeed, effective instructional leadership of leaders in a school is important to teachers' performance and students' academic achievement (Strauss, 2013). One of the supervisory responsibilities by heads of schools is to find out what goes on in the classroom. Observation in the classroom should be done from time to time and where necessary teachers should be advised on their strengths and weaknesses. When heads of school praise teachers it usually makes teachers feel valued and this is likely to inspire teachers to work for longer hours. To provide helpful advice to teachers should lead to an improvement in their performance. Evaluation of teachers is one of the responsibilities of heads of schools. School leaders should be able to communicate in an excellent way. They should continually recap the school's vision, mission and values to teachers, parents and students and their core mandate as a school which is to provide teaching and learning in a safe environment. School heads should encourage teachers that with determination, they can survive in difficult times. In fact, it is prudent that they listen to all stakeholders in the running of their schools. It is also relevant for school leaders to follow the theory of shared leadership which involves other stakeholders in the field of education. Ideas and views of those in the school environment might help improve academic performance if heeded

to. Their skills in communication could instill confidence, motivation and skills among students and teachers, leading to a positive impact on students' performance. Collaboration among teachers should characterize a school's leadership style of leading. Meeting of teachers should help promote a safe place of exchanging knowledge and experiences in relation to teaching and learning. Hence, this prevents the idea of isolation among teachers on students' performance, which is a major issue on the part of academic staff. In addition to leading to a collaborative style among, it is expedient that school leaders should be easily approachable by all, lead with a positive mindset, seek to be a role model for students to emulate and examine a process that improves teaching and learning as well as provision of staff professional development. To be successful in leadership requires that one should have excellent planning and observational skills in addition to expertise in research and evaluation for teachers and students (Geshel, 2017)

2.5.2 Teachers' Characteristics

School improvement depends on teacher characteristics. Teachers are those who bring about development and improvement in students' knowledge and serve as agents of change (Wallace, 2011). Effective teaching is a key feature of, among other interests of policy makers in education, among teachers and parents. This is achieved through terminal and final examinations. On the one hand, teachers have a feeling that students who fail at exams are taught by inefficient teachers and those who are able to pass their exams are tutored by efficient teachers. Research in this field suggests that teachers' effectiveness has a relationship on students' academic accomplishment (Afe, 2001). It stands to reason that teachers serve as facilitators in the course of their teaching and

learning activities. They seek to bring an interpretation to students on the subject they handle bearing in mind the theory and practice components of the subject. As Taal (1996) noted the teacher is gradually the focus of attention due to the role they play in delivering quality education to students. This implies that teachers are considered as persons having the ability to change the academic performance of students. They are worthy of praise and also given perks when they perform well. In Botswana, the government has made it a point to reward both students and teachers who excel in their external examination and these include the receiving of gifts and fully paid trips to outside the country. Teachers are responsible for imparting knowledge and skills in students in addition to encouraging them. This is incorporated into the employment contract and clearly spelt out in their responsibilities as teachers. Teachers are known to help students develop their potential abilities in addition to developing the knowledge, skills and attitudes. At the beginning of a student's career, it is the duty of a teacher to nurture a student's primary skills required to help further development in academic work. Teachers serve as guides for students in achieving their academic goal in life. As a result, students should be prepared to face situations that may occur in the course of their education. In this era, according to Bubblews (2013, p. 3) teachers should "develop learners instead of teaching them, who help their pupils to become independent (learning to learn), who provide students with motivation and interest for lifelong learning and urge them to become autonomous learners". Teachers make use of a number of techniques in order to improve academic goals. This involves evaluation, assessment and provision for students who require special attention as far as academic work is concerned.

To sum it up, the characteristics of teachers are multifaceted and this should urge all teachers who feel they lack certain traits to fully acknowledge them and find innovative ways of improving on those traits.

2.5.3 Students' Characteristics that Affect Academic Performance

A lot of authorities posit that the following factors (Type or nature of the School; Students' Learning Preferences; Class attendance and academic performance; Gender factors and Entry qualifications and prerequisites) constitute students characteristics that affect academic performance (e.g. Kwesiga, 2002 and Sentamu, 2003; Crosne and Elder2004; Harb and El-Shaarawi, 2006; Reid, 1995; Pashler, McDaniel, Rohrer, and Bjork, 2008; Devadoss and Foltz, 1996; Durden and Ellis, 1995; Romer, 1993; Park and Kerr, 1990; Schmidt, 1983). Based (Rodgers and Rodgers, 2003; Durden and Ellis, 1995).

These factors are discussed below:

2.5.4. Type or Nature of School

How a school environment is and its teachers' expectation from its students are strong factors that influence a student's performance (Considine and Zappala, 2002 as cited by Sparkles, 1999). A teacher who works in a poor school setting might run into the lack of basic facilities which is required for academic work to take place and when students perceive that teachers are expecting a low performance on their part, it is likely to result in that. Kwesiga (2012) is of the view that the performance of students is often influenced by studies in the school and the number of facilities available which

corresponds to the rank of the school. This commensurate the performance and accomplishment of its students.

Sentamu (2003) underscore that a school's influence is based on its educational process in organizing its content, and also its teaching and learning process, resulting in its overall evaluation. Students who attend top-rated schools are normally expected to perform better because of the school they attend, which is seen to be well resourced. This is as a result of its resources which school authorities appropriate and also the available funds allocated for such activities, hence leading to a better performance of their students. Crosne and Elder (2004) point out that school ownership, provision of facilities and resources available as important structural components of a school setting. To conclude, the above authors are in support of what constitutes the nature of a school if it is to succeed or otherwise and paying heed to these factors are likely to improve academic performance of a school.

2.5.5. Students' Learning Preferences

Reid (1995) has defined learning preference as the "natural, habitual and preferred way" of grasping new information. Research by Harb and El-Shaarawi (2006) indicates that there is a link between learning preference of students and their teacher's teaching style. Every individual has a way of understanding what is taught effectively due to difference in them. In order to encourage learning preferences among students, teachers should be able to analyse student preference with regard to learning and adapt to that preference (Pashler, McDaniel, Rohrer, and Bjork, 2008).

Still, Omrod (2008) has come to understand that information which is put across in the form of words are better understood by some students, while other students are able to grasp such information in pictorial form. Teachers who make use of only one learning style are likely to affect the way students understand what is presented due to difference in students learning styles thereby affecting their performance academically. Previous research by Felder (1993) has argued that there is a link between a learner's preference and an instructor teaching style which should result in a better memory and comprehension. Moreover, learning preferences has been widely researched in the field of education despite its lack of evidence to support the claims.

2.5.6. Class Attendance and Academic Performance

Among researchers in this field, Romer (1993) is one of the authors whose work has been cited largely and it sought to examine the link between student attendance and examination performance. Many factors contribute to the fall in students' performance over the past 15 years. Key reasons given include "assessment pressures, poor delivery of lectures, timing of lectures, and work commitments" (Newman-Ford, Lloyd and Thomas, 2009). The critical question worth asking is, "does absenteeism affects students' academic performance?" Meanwhile, several authors have argued that students who miss class might perform badly (Devadoss and Foltz, 1996; Durden and Ellis, 1995; Romer, 1993; Park and Kerr, 1990; Schmidt, 1983). As a result of this finding, stakeholders in the education sector are of the view that class attendance should be always checked.

However, evidence suggests a strong relationship between attendance and academic performance, and it seems that none of those studies mentioned beforehand specifically show a causal effect. The lack of longitudinal studies which focuses on students' characteristics is a main setback to the usefulness of these findings (Rodgers and Rodgers, 2003). It is clear that class attendance and academic performance have a strong relationship and the absence of one is possible to impact on the other.

2.5.7 Gender Factors

The impact of age as against gender on academic performance has been the focus of some researchers which has led to varied conclusions on the subject. Differences in what researchers have focused in this field might vary based on the subject of study of a student, age and gender. It has been found that men perform better than women in some circumstances, whereas women also do likewise in certain fields of study (Haist, Wilson, Elam, Blue, and Fosson, 2000).

In contrast, a study by Borde (1998) failed to provide evidence for academic performance as influenced by gender. For an analysis of 2,000,000 students about to graduate, Woodfield and Earl-Novell (2006) found that females performed better than males due to the fact that they are meticulous and may not absent themselves from class. From the above views of experts in the field, it is evident that both boys and girls may have subjects they excel at and should be encouraged to pursue their passion.

2.5.8 Entry Qualifications and Prerequisites

Anecdotal evidence suggests that senior high school (SHS) students will often provide their Basic Education Certificate Examination result slip provided by West Africa Examination Council (WAEC) plus the placement form provided by The Computerized School Selection and Placement System (CSSPS). Previously, a JHS student graduate should have a grade from six to thirty to enable all students who chose the subject get the opportunity to read the programme. Currently, the Government of Ghana policy for Free SHS for all students has made it possible for students with even aggregate forty and above get access to senior high education.

According to Mlambo (2011) learning is a cumulative process, that is, if a student is admitted with a good grade into the school, that student will be ready for the course material than the one admitted with a weak grade. School authorities should investigate the background of students in order to know their previous performance.

2.6 Parents or Guardians Factors

Studies have shown that when parents participate in their ward's education, it has a positive effect on their academic performance and the school's success (Halsey, 2004; Christie, 2005). Involvement of parents in their ward's education improves the morale of teachers and thereby may create a cordial relationship between the school and the community. The important aspect of it is that of parents involving themselves in their ward's education. There are several things that parents or guardians can do to help their ward's education. The purpose of the involvement promotes positive behaviour between students, checking up on students' work, encouraging contributions from each parent

towards their ward's education, helping students to do their assignments, tutoring, supervision of afternoon study periods, coaching sports and motivational talks. Provided that parents are aware that their opinions are factored in their ward education, they will be interested in providing such assistance to their ward education. The wish of a parent is to have a child who becomes a good citizen. An aspiration of a parent is to see their ward succeed in their education. Moreover, developing countries that have a high percentage of unemployment rate usually have parents focusing on their ward's education which they believe with result in career choices and lucrative jobs.

2.7. Home Conditions of Students that Affect Academic Performance

Parents financial status and their education level, occupation, and their academic status affect their wards' academic performance. Many studies corroborate that the academic achievement of students is possibly based on parents socio-economic condition. To this end, students from higher socio-economic background will outperform their counterparts from low socio-economic background together with the work they do and their income level (Jeynes, 2002).

Researchers such as Graetz (1995), Considine and Zappala (2002), Eamon (2005) and Pedrosa et al. (2006) have undertaken research on academic achievement, and as such it is not very surprising to note that the socio-economic status is a key element when making projections for academic performance. In fact, Graetz (1995) conducted a study of socio-economic research with parents of students and came to conclusion that the social status might have a great impact on students' academic performance; thereby

creating imbalance between students and their success which is possible through their socio-economic status.

Two authors (Considine and Zappala, 2002 and Graetz, 1995) are of similar views that social and economic demerit in the academic performance of students when parents or guardians have social, economic and educational advantage may achieve a higher success rate in the future. However, it has come to the notice of parents who make available sufficient psychological and emotional support to their children by providing a sound education and learning environment which result in confidence and the improvement of the requisite skills to succeed.

On the contrary, students from poor homes are likely to perform poorly based on the socio-economic and educational background of parents, and the opposite is true for those from wealthy families (Pedrosa et al., 2006). This occurrence is called educational elasticity. It is noteworthy that the criteria in the classification of socio-economic standard in different nations vary in terms of their norms and values. The checklist for low socio-economic status are made known to developing nations and those from high socio-economic principles are different as well.

The income of a family for a month or year and their expenditure also put a great effect on the learning opportunities accessible to the younger generation and their opportunity to achieve success. A parent's income or social status could have a positive effect on a student's test score. It is presumed that students learning outcome and educational performance have a relationship with the standard and type of educational institution a

student may find him or herself with an education setting. The educational atmosphere of the school one attends sets the limits of students' learning outcomes.

2.8. Strategies that can be put in Place to Improve Students' Academic

Performance

Gagne (1990) has observed that learning occurs when a learners' performance has changed repeatedly. This is what Gagne considers to be learning. Ehiametor (1990) is of the view that after learning, assignments to test students' knowledge on what has been taught follows. For instance, when a teacher is teaching in class, it seems that students are able to follow through, but as soon as similar problems are given students are likely to find it difficult to solve such problems. Teachers are able to gauge the students' weakness and prepare towards such weakness in helping students overcome those areas in his or her subject area.

A significant factor in learning as stated by Ehiametor (1990) and Farrant (1992) is the learning environment. A conducive atmosphere is what makes learning more effective. Moore (1998) opines that teachers who are zealous are likely to produce top students who can achieve more in their academic pursuit. Farrant (1982) argues that students perform better when their fellow students teach them and this usually underscores what the teacher has already taught in class. This leads to better students exercising their teaching ability. Hence, this suggests how learning should take place in an effective teaching atmosphere. The importance of teaching and how it is handled in the classroom is very important. This is shown through the UK Government's effort in the ways

teachers teach in a more prescribed manner such as literacy, numeracy, among others (Mortimore et al., 1998).

Mortimore et al. (1998) has noticed that teachers who spent more time in routine exercises with individual students use low order questioning strategies and those who go for discussion in class often use the high order questioning strategies. Moreover, the factors that make effective teaching among students include "structured sessions, intellectually challenging teaching, a work oriented environment, communication between teachers and pupils" (Mortimore et al., 1998). The following are considered to be good traits of teachers as stated by OFSTED (1995): good subject knowledge, good questioning skills, an emphasis upon instructions, a balance of grouping strategies, clear objectives, good time management, effective planning, good classroom organization, and effective use of other adults in the classroom.

Learning by doing involves students' active participation in a particular subject in the field of education (Zapalska et al. 2002). This result in the student's ability to learn on their own and they should adapt to changes in the learning environment. As a result, teachers could encourage student learning on their own. Teachers should give learners the ability to contribute at certain when they are teaching in order for learners to discover things by themselves and contribute meaningfully in class. In this vein, "teachers must therefore act as consultants, catalysts, arbitrators, troubleshooters, confidants, fellow learners and sign posters, managers, communicators and facilitators in the classroom" (Mordedzi, 1999). As described by Farrant (1980), effective teaching considers how a teacher teaches with respect to students, the curriculum, resources and teaching methods. It involves the requisite knowledge of child and adult development

and teaching skills in order to structure and deliver effectively in his teaching. From the above views of different authors, it is clear what teachers and students should do in order to improve the academic performance which has effects on both students and schools.



CHAPTER THREE

METHODOLOGY

3.1 Overview

This section of the study discusses the research design, the data collection instruments, the study population, the sample size, the sampling techniques, data analysis and presentation procedures, criteria reliability and validity, reliability and validity of results, and finally ethical consideration.

3.2 Research Design

A research design gives details of the basic structure of the research as it offers the procedural framework for the conduct of the research in line with the researcher's preferences, philosophy, and ideas (Amedahe, 2004). According to Saunders, Lewis, and Thornhill, (2009) the types of design or plan for a research; is exploratory, explanatory, and descriptive. Saunders et al. (2009) maintains that exploratory research is adopted on a research field that have not yet been discovered in previous researches and strives to generate hypotheses for more research. Explanatory research analyses the causes and relationships and make efforts to identify patterns related to the subject matter or phenomena studied.

Exploratory research is thus explained as finding new insights and evaluating occurrences in a new light (Robson, 2002) whereas explanatory study centres on enlightening relationships among variables (Saunders, et al., 2007). In descriptive research, a subject matter or phenomena is described and documented. It is not

necessary to use all these three in a particular research as the aim of the study was not to develop new propositions for future research (Yin, 1994) or an explanation to decide the impact that a variable has over another though the problem was considered structured (Emory and Cooper, 1991).

3.3 Research Approach

3.3.1 Qualitative Approach

Hollensen (2007) describes a qualitative research of been able to provide "a holistic view of a research problem by integrating a larger number of variables, but asking only a few respondents."

Qualitative in simplest form deals with words and for this purpose provides opportunity to rely on observation and fieldwork, which help to understand a reality or to understand an object (Bryman and Bell, 2007) under study. The qualitative approach helps to better understand social phenomena as it provides opportunity to deeply understand and investigate the phenomena or topic (Creswell, 2007), in a totality which in this study refers to the factors that affect students' academic performance of visual arts education in Dormaa Senior High School and the strategies that need to be put in place to enhance the Visual Arts Program in the school.

3.3.2 Qualifications of Qualitative Approach

With the nature of the research questions (being "what"), the need for control over behavioural events, and the descriptive nature of this study coupled with adoption of case study, as specified by research methods experts (e.g. Yin, 2003b; Ghauri and Gronhaug, 2005; Baverstam and Larsson, 2009), qualitative becomes the required

approach since according to Holme and Solvang (1997), within qualitative method, the researcher's understanding and interpretation of the phenomenon under study that stands in the centre. Secondly, a qualitative approach was chosen because it was considered useful for the study aimed to obtain a comprehensive information on factors that affect students' academic performance of visual arts education in Dormaa Senior High School and the strategies that need to be put in place to enhance Visual Arts education in the school.

Thirdly, the researcher intended to use qualitative research to test theories inductively without any intention of quantifying findings. Hence, with the adoption of qualitative approach, the researcher described certain aspects of the phenomenon relying on the researcher's interpretation of the actual situation (Cormack, 1991; Leach, 1990 as cited in Charoenruk, 2006; Holmen and Solvang, 2001).

Lastly, in defending why qualitative approach was chosen, the researcher was informed by empirical study that the gathering of data that involved larger, more respondents' samples and numerical calculation of results (Wiid and Diggines, 2009), or findings. As researching into the factors that affect academic performance of Visual Arts Students of Dormaa Senior High School in Brong Ahafo Region in Ghana does not need large sample size selection, because information could be collected effectively by contacting forty(40) key respondents in Dormaa Senior High School being the case-studied school.

3.4. Research Methods

Research strategy, according to Saunders et al, (2007), is defined as the general strategy of how a researcher will go about providing answers towards the research questions. Remenyi, Williams, Money and Swartz (2003) defined research strategy as providing broad direction of the research as well as the process by which the research is conducted. Considering the topic under discussion, the researcher sought to adopt a case study and descriptive research method as the appropriate strategies on the subject. The research is based on factors that affect students' academic performance of visual arts education in Dormaa Senior High School and the strategies that will enhance the Visual Arts programme in the school. The following section briefly describes case study strategy and justifies its preference in this study as well as description.

3.4.1 Case Study

Creswell (2003) defines case study as "a qualitative approach in which the investigator explores a bounded system (a case) or multiple bounded systems (cases) over time, through detailed, in-depth data collection involving multiple sources of information (such as observations, interviews, audiovisual material, documents and reports), and reports a case description and case-based themes". A case study becomes a suitable strategy when the researcher decides to focus on one or few research units with the purposes of getting deeper insights into events, relations and processes (Denscombe, 2000). It is stated that when finding answers to "what" research questions, when the researcher have little control over events, and when the focus is on a phenomenon in a

real-life context, case study becomes the preferred strategy (Yin, 2003b; Ghauri and Gronhaug, 2005).

With the focus of the research being to gain deeper understanding of the factors that affect academic performance of Visual Arts Students in Dormaa Senior High School, and giving the nature of the research questions, the need for control over behavioural events, and the descriptive nature of the thesis topic as specified by Yin (2003b), and Ghauri and Gronhaug, (2005), all these qualifications informed the researcher's decision to conduct a single case.

• Justifications for the adoption of Case Study method

There are three qualifications recommended by Yin (2003b), as regards the adoption of case study method that is forming the basis for its preference in this present study. In the present thesis, case study was preferred and justified. The usage of "what" questions are evidenced in the research questions below:

- i. What are the specific factors that underpin low academic performance of Visual Arts Students in Dormaa Senior High School?
- ii. What are the strategies that need to be adopted to get the academic performance of Visual Arts Students in Dormaa Senior High School enhanced?

The second condition as identified in Yin (2003b), was the degree of control the researcher has over the actual behaviour of respondents. In this thesis work, the researcher had no control over the behaviour of selected students and the teachers who

were chosen as participants for the study. Furthermore, there was no possibility of manipulating the behaviour of all the respondents selected from the school, and regarding the subject matter, factors that affect students academic performance of visual arts education in Dormaa Senior High School and the strategies that need to be put in place to enhance Visual Arts Programme in the school. This means that all these events were beyond the control of the researcher.

The last condition is the degree of focus on contemporary or historical events. Ideally, the phenomenon under investigation was contemporary one, that is, factors that affect students' academic performance of visual arts education in Dormaa Senior High School and the strategies that need to be put in place to enhance the Visual Arts program in the school. This made the third condition for selecting case study research also justified, and preferred in this study.

As a type of research strategy, case study makes use of several methods to generate knowledge and ideas. Triangulation has been described as the essence of adopting case study and is considered as imperative to ensure validity as quality criteria in a case study (Johansson, 2003), method. The meaning of triangulation, is to "... use multiple research methods across multiple time periods." To constitute triangulation, the researcher used three different common methods, which were mentioned by Woodside, (2010) to include:

 Direct observation within Dormaa Senior High School being the scope of the study.

- 2. Probing by asking participants during the focus group discussions and the explanations and interpretations of the interviews.
- 3. Analysing written documents such as attendance register, terminal reports, placement forms for admissions, and WASSCE results analysis, hence the researcher used triangulation being generated from case study method.

The sampling selection was done through judgemental, in order to ensure that all the students and teachers chosen as respondents had the characteristics of fitting into the profile and every interviewee was available, willing and equipped to provide the needed answers to the questions posed during the focus group discussion and interview session. In this wise, it needs to be stressed that only students, offering visual arts and their teachers (in GKA, Leatherwork and Graphic Design) as the elements to be studied. All these respondents were located through the researcher's network of contacts, hence negotiating access became easy.

3.4.2 Descriptive Method

The most appropriate strategy for conducting this study was descriptive research methodology which was adopted to design and collect data for analysis. It is instructive to note that when touching on the factors that affect the academic performance of visual arts students in senior high school in the Brong Ahafo Region in Ghana, it is assumed that the researcher already had some knowledge and information about the chosen area. So what followed was to do a comprehensive description of the subject matter or the phenomenon. In adopting descriptive research design for this study, data collection took

the form of focus group discussions and Unstructured Interviews. These were used to collect the necessary data from the students and teachers of Visual Arts Department of Dormaa Senior High School, being the case-study school.

The blueprint of this study was to do a detailed description of the phenomena as it exists in the majority of senior high schools in Ghana. The last but not the least reason for conducting descriptive research was to define the features of the phenomenon (Zikmund, 2000), which for this situation is referred to the critical factors that affect students academic performance of Visual Arts education in Dormaa Senior High School and the strategies that need to be put in place to enhance Visual Arts programme in the school.

3.5 Population and Sampling

3.5.1 Population

To gather data for this study, the researcher had to do that from a total population of one hundred and thirty-two, and more specifically from an identical sample. The term population, in this case is defined as "the universe of units from which the sample was to be selected" (Ghauri and Grønhaug, 2005). It is also described as the "first step" in adopting sampling method to select a particular sample in a group of units which is termed as population (Bryman and Bell, 2011).

The population for this study consisted the students pursuing the Visual Arts Education in Dormaa Senior High School and the past students from 2014-2016 year group. All the three classes in the department and the past students of the three year groups (2014-

2016) are considered for the study. The researcher believed that students in these classes might have knowledge about WASSCE and that they are also familiar with factors affecting their academic performance. In addition are the visual arts teachers.

3.5.2 Sample Selection

It has to be noted that to include every individual participant in the population for a study can be considered so close to be impossible and impractical. Also, it should not be assumed that a census would necessarily and practically provide more useful results than gathering data from a sample which represents an entire population.

That is to say, the sample selection is a very significant step in the research process, particularly when the researcher is constrained in terms of time and the results to be collected from the data are needed quickly (Saunders *et al.*, 2009). A sample is "the segment of the population that is selected for investigation" (Ghauri and Grønhaug, 2005).

3.5.3 Sample Size

A sample size of forty(40) respondents were purposively and conveniently selected out of one hundred and thirty-two(132) to help in eliciting responses for this study. To get all the forty (40) respondents, a quota of nine (9) Visual Arts students from each year group was selected from all the three previous year batches (that is, 2014, 2015 and 2016) and the current students, making thirty-six (36) and a total of four (4) teachers were selected to be interviewed.

This is illustrated in Table 3.1:

Table 3.1: Number of respondents

Year Batch	Number of respondents selected
2014	Nine (9) VAS
2015	Nine (9) VAS
2016	Nine (9) VAS
9 Current students	With three (3) selected from each year
(2017, 2018, 2019)	class=(9)
Total number of students who participated = 36	

Author's construct, 2017

Hence, this brought to the fore the criteria which were used to purposively sample participants to participate in this study, which included the following:

- Visual Arts students (both past and current: 2017; and current form 3 and form 2 students).
- Students (past and current with the time and willingness to accept to participate in this study specifically in the focus group discussions and the unstructured interviews).

With regards to the population of this study, interview as a data collection method was adopted. Four teachers were also selected to be interviewed on the subject matter, "factors that affect students' academic performance of visual arts education".

It is instructive to note that all these categories of respondents were chosen for each of the data collection techniques because they were considered to be the right people needed to give the right information on the subject. Holme and Solvang (1997), posit that it is of great importance to find the respondents to obtain the right information from them.

3.5.4 Sampling Method: Non-probability Sampling

In adopting sampling selection process, there are two main alternatives available. They are: the random sampling, which deals with the researcher ensuring the possibility to generalize the empirical findings, and the non-random sampling. Saunders *et al.*, (2007) stresses that probability sampling also known as random sampling is most commonly associated with survey-based research strategies. Conversely, "for non-probability sampling, it is impossible to answer research questions or to address objectives that require statistical inferences about the characteristics of the population." The authors view it as a type of stratified sampling sometimes referred to as "judgemental or purposive sampling" or expert choice (Saunders *et al.*, 2007).

In the latter process, the researcher decided based on her own subjective judgement which respondents should be included in the study. Previous idea of the sampling could lead to some representatives to be favoured before other representatives. Due to time and other resource constraints in this master's project work, convenience sampling of the key respondents (that is, Visual Arts students and their teachers) was done. Through a self-selection process, the representatives involved are voluntarily participating in the

study and usually noticed earlier in time (Johansson, 1993). The researcher arrived at a conclusion that a non-probability sampling method remained suitably qualified for the study, since a smaller number of Visual Arts Students, thirty-six (36) and their four (4) teachers were available. Snowballing or referral technique was adopted to get the past students (who have completed school for the past three years) participate in the focus group discussions.

3.6 Sources of Data

It is widely known that researchers in their quests of carrying out studies are able to gather data through several means such as observation, experiment, interview, survey, focus groups, panel discussions, etc. It needs however, maintain that the choice of data collection is dependent heavily upon the type of data needed (Ghauri and Grønhaug, 2005) to be collected. It is known as the rule of thumb that any form of data can be classified into primary and secondary data (Emory and Cooper, 1991).

According to Hollensen (2007), primary data are simply, "information that is collected first-hand, generated by original research tailor-made to answer specific current research questions".

For the purposes of this study, the data sources were primary data ones that were gathered directly from the field using instrumentations such as focus group discussions and unstructured interviews, whereas, secondary source of information simply comprised published articles from peer reviewed journals, internet sources such as dogpile.com, etc.

3.7 Data Collection Methods

3.7.1 Focus Group Discussions

Focus group discussions and unstructured or in-depth interviews were used to collect data for this study. Morgan (1996) defines focus group discussions "as a research method that collects data through group interactions on a topic determined by the investigator". According to Zikmund and Babin (2010) focus group discussions comprise unstructured, free flow interview with a group of participants usually between six and ten. Including significant narrative data which was found in the focus group discussions improved the study's validity and reliability standards (Halcomb, Gholizadeh, DiGiacomo, Phillips, and Davidson, 2007). Conducting more than a focus group discussion was required to provide sufficient exploration (Shoqirat, 2009) and was likely to lead to the increase in the reliability of the data to be gathered by detecting the agreement across differing focus groups (Morgan, 1997). For the purpose of this study, issues that emerged from one focus group discussion triggered the discussion in successive focus groups.

Following the position of Zikmund and Babin (2010) regarding the number to be chosen for a focus group discussion, a total of thirty-six (36) respondents – nine (9) past students were selected from each past year; 2014, 2015 and 2016 respectively. Another (9) nine respondents being the current Visual Arts students were selected (comprising three students selected from each class).

All these respondents were selected because they remained the right individuals and in the best position to provide the right responses or information needed to carry out this study (Holme and Solvang, 1997). Practically, the students who were selected for the unstructured interview from each year completed batch together with the current students were all considered the most suitable participants for this study because they remain individuals most likely to have greater understanding when it comes to unearthing the factors that affect students' academic performance of visual arts education in Dormaa Senior High School. On the other hand, teachers were also considered right respondents because they impact knowledge being acquired by the students. Hence, they too have a wealth of information concerning factors that affect students' academic performance of visual arts education in Dormaa Senior High School.

In each year's batch, two forms of data were gathered and compared with one another for the purposes of gaining the most preferred data to be gathered from each participant. The focus group discussions were based on the factors that affect students' academic performance of visual arts education in Dormaa Senior High School.

With the researcher acting as the moderator, a series of sub-questions derived from the three main research questions were posed to the participants. The structure of the questions for the focus groups was as follows:

- i. What are the factors that result in high students` academic performance of visual arts students in Dormaa Senior High School?.
- ii. What are the specific factors that underpin low academic performance of Visual Arts students in Dormaa Senior High School?
- iii. What are the strategies that can be put in place to enhance academic performance of visual art students in Dormaa Senior High School?.

3.7.2 Unstructured Interview Method

After the focus group discussions, the same participants were used to conduct a one-onone interview.

These interviews hinged on the same objective, but questions were posed a little bit different in a form similar to the former structure. These interviews were used to establish if the responses from the focus group discussions departed significantly from the One-On-One Interview. They were also used to check if groupthink had an influence on the information that were gathered from the first two focus group discussions. The challenge associated with this method was to ensure that the differences in the responses from the focus groups and the One On One Interview, if any, were attributable to groupthink. The actual limitation concerned with how potential sources of falsity and inaccuracies were to be controlled, relating to variables grouping and individual responses in all the focus group discussions. To deal with this limitation of spuriousness associated with the responses given by the respondents a brief period of thirty minutes existed between the focus group discussions and the One-On-One Interview.

Before the focus group discussions and the unstructured interview, permission was sought from participants to do audio recording for responses from them using a digital recorder. During the discussions and unstructured interview, the researcher adopted brief note-taking to augment the digital recording of the responses, which was followed up by detailed notes that were taken right after each interview. This was to ensure the capturing of as much detail as possible from each interview.

From a hermeneutical standpoint, the one-on-one interviews comprise text, but not statistics (Huei-Wen and Huei-Fu, 2017). This study was based on hermeneutics because it provided a support during the analysis stage of the chosen theme (that is, the factors that affect academic performance of Visual Arts Students) by gathering data and literature. The study primarily concerned with the viewpoints of Visual Arts Students and their teachers in the school.

3.7.3 Pilot Study

When organizing the focus group discussions for the study, the researcher realized that little information could be obtained on the factors that affect students' academic performance of visual arts education in Dormaa Senior High School.

Thus, it was difficult to know which question should be posed during focus group discussions and how to structure them in order to provide answers to the research questions. A survey questionnaire was designed and submitted to the Visual Arts students and teachers selected as the respondents for the study. The researcher realized then that some mistakes might have been made. Through this pilot study, the researcher realized that certain questions were unnecessary while some others were good. Afterwards, the whole structure was modified to get improved set of questions posed to help answer the research questions.

3.8 Data Analysis

The data in this study were analysed using a standard procedure. It is contended that thematic analysis being the main method of analysis in this study is seen as "essentialist or realist" method (Braun and Clark, 2006), following the tenets of the study, which were based on description and reporting on experiences and reality of students being participants in the focus group discussions, whereas, teachers also selected as participants in the interviews. It was against this premise that thematic analysis became applicable to the study as it examined both students and teachers' views in relation to the factors that affect students' academic performance of visual arts education in Dormaa Senior High School. Thus, it provided an opportunity for the study to have a reflection of current realities and to which led to the exploration of the surface of this reality that emanated from the perspective of the respondents.

By using thematic analysis, the researcher painstakingly examined all the questions posed to participants during the focus group discussions and unstructured interviews. The data were analysed in the three steps of thematic analysis, which were further grouped into categorizations (i.e. dividing findings into groupings). Once the themes and the categorizations were finalized, explanation building was adopted to examine within the context of each question posed in the interview guide and schedule.

The first objective, that is, the general factors that affect students' academic performance of Visual Arts Education in Dormaa Senior High School, comprised five categorizations – School based factors; Teachers' Factors; Student's characteristics/demographics; Parent's/ guardian socio-economic condition and Home Conditions Or Parental Support.

The second objective being specific factors that underpin low academic performance also comprise five categories — School based factors / School leadership; Teachers' Factors; Student's characteristics/ demographics; Parent's/ guardian socio-economic condition and Home Conditions Or Parental Support.

The third objective being the strategies that need to be adopted to get the academic performance of Visual Arts students in Dormaa Senior High School be enhanced. Also, it comprised of two categorizations: the role to be played by the leadership of Dormaa Senior High School in adopting strategies to enhance academic performance of Visual Arts students in the Visual Arts Department and specific roles to be played by teachers in adopting strategies to enhance academic performance of Visual Arts students. The specific role to be played by the students to enhance academic performance, specific role to be played by Dormaa Ahenkro community to enhance academic performance of visual arts students and the specific role to be played by old students to enhance academic performance of Visual Arts students.

All the manuscript obtained from each of the focus group discussions and the unstructured interview was transcribed exactly into separate identified folders. The digitally recorded focus group discussions and one on one interview were re-played many times to ensure that the data obtained were understood adequately by the researcher.

The researcher read through each manuscript repeatedly, to become familiarized with the complete data gathered (hence making it deductive analysis). Then, line by line a search for the manuscript was carried out to have central themes scanned through. These also included repeated ideas or statements (Theme) "which implied something" (Brunard, 1991) about the topic. This was followed up with making brief notes about each manuscript using different colors for different themes (e.g. selected ideas that repeat have the same colour that identify them).

Once again, each manuscript was re-read to ascertain that common themes were really common in all manuscripts. Also, by so doing irrelevant materials which were referred to as "dross" (Brunard, 1991) were identified and deleted from the analysis. To ensure the credibility of the findings and results in this qualitative analysis, the researcher all the time checked the manuscripts to eliminate inaccuracies and ambiguities.

3.9 Quality Criteria: Reliability and Validity

3.9.1 Reliability

Bryman and Bell (2011), maintain that the goal of reliability as criteria for assessing quality standards is to minimize errors and provide stable results of the data to be collected. The researcher took into consideration that the right sample was selected from the pool of respondents since there was the need to have the right respondents who have knowledge of the factors that affect students' academic performance of visual arts education in Dormaa Senior High School.

Also, more information was obtained from secondary data through journal articles and textbooks. All the journal peer reviewed articles and sources for textbooks obtained from the KNUST were scrutinized to conform to reliability and credibility status. For

the researcher to be sure about the reliability of the articles and books, all of them were read and compared to establish consistency before being used.

3.9.2 Validity

It is established that the areas of validity for qualitative study comprise descriptive, interpretive, theoretical and generalizability (Ghauri and Gronhaug, 2005). Descriptive validity deals with to what degree does the descriptions actually hold true. This was achieved to an acceptable degree through the data collection. The researcher having both taken extensive notes and putting most of the words that came from the interview on tape recorder, this allowed the author believe the interpretations made in the study to be valid. The theories and models for extant studies used were selected based on the fact that emerged legitimate and credible sources, hence considered as highly applicable to this study.

In the main, it is instructive to note that the interview subjects or those chosen as key respondents were all competent with the subject under discussion, having been in the key position to contribute to the provision of information regarding factors that affect academic performance of Visual Arts students in the selected Senior High School. This suggests that the findings of this study hold true and the study theoretically valid.

3.9.3 Reliability and validity of the Study's Results

To enhance the reliability and validity of the results, these activities were done by the researcher:

First, objectives had a correlation with the research questions posed with enough literature to support both the objectives and the research questions. Second, it was imperative to have the right respondents selected for the interview process. These were the ones with sufficient knowledge when it comes to the subject, factors that affect academic performance of Visual Arts students in the case studied school.

Third, before the interviews were conducted, several meetings took place to establish a rapport with the respondents. To the researcher, that was hopefully made to make the respondents feel more comfortable when giving out the information.

Lastly, as the researcher was present during the unstructured interview, the researcher had the opportunity to get the interpretations confirmed. Summary of the interview findings was sent to the respondents to make sure nothing was misinterpreted. These efforts were meant to strengthen the reliability of the results. In essence, the responses obtained from the different respondents were found to be similar, something that made the researcher trust the results of the study. This also indicated that the results can be related to a certain extent.

WUSANE

CHAPTER FOUR

PRESENTATION AND DISCUSSION OF FINDINGS

4.1 Overview

This chapter presents the analysis and the results obtained from the field in the focus group discussions and one-on-one interviews. The responses received from the field addressed the three objectives, which comprise general factors that affect academic performance of Visual Arts Students in Dormaa Senior High School. Specific factors that contribute to low academic performance of Visual Arts students in the school and the strategies that need to be put in place to enhance effective academic performance of Visual Arts students. This study therefore sought to find out the causes of the decline in students' academic performance and to propose strategies that will enhance students' academic performance.

4.2 Objective 1: Factors that Result in High Academic Performance of Visual Arts Students in Dormaa Senior High School

Five categories were obtained under this objective which included: School based factors/School leadership; Teachers' Factors; Student's characteristics/demographics; Parent's/ guardian socio-economic condition and Home Conditions Or Parental Support.

All these categories summed together indicate the extent to which participants contributed to the discussion which also revealed that they had worth of information regarding factors affecting academic performance of Visual Arts students in Dormaa Senior High School.

4. 2.1: School based Factors / School Leadership

The prevailing factors inherent in the Dormaa Senior High School Visual Arts students are capable of influencing their academic performance. This stems from a number of teachers with high academic qualification, teaching and learning materials, judicious use of contact hours by teachers, medium of instructions in the school, and tuition trend.

The school authorities can provide guidance and counselling to parents to create a positive home environment for improvement in the students' quality of work. The direction from the school authorities sometimes influences performance of their students. The guidance from the school authorities indirectly influence the students' performance.

Similarly, the school's leadership (which is headed by the Headmistress and the various Heads of Department appointed by her) supervise academic progress of the various departments of the school, organize workshops for the students to help improve their academic performance. Moreover, it is its duty to ensure there is discipline to promote a peaceful environment for teaching and learning to thrive; evaluate or appraise the performance of teachers at the end of every academic year.

Evidently, it was revealed that it is the overall Head of the school who recommends the appointment of any teacher through the consultation of Head of Department (HOD) but it is the same Head of Department who selects teachers to teach the subjects that fall within their areas of specialization. Also, the Heads of Departments are delegated by the overall Head of the School to monitor the teachers under their supervision e.g. Monitoring to ensure consistency regarding the content of the syllabus and the actual

content being delivered to the students, both Teachers and students' punctuality, absenteeism and attitude towards work are all monitored by the Head of Department who also report to the overall Head of the School on these. Teachers as part of the school's leadership also report to their Heads of Departments regarding the students they teach and their academic performance. Additional teaching and learning which is also known as extra classes is also sanctioned by the overall Head of School through approval of the Parents Teachers Association.

Lastly, disciplinary actions are taken by the overall Head of School against teachers and students who sometimes do not conform to the rules set by the school. These punitive actions go to the extent of serving as a deterrent to others.

In all, three past students from 2014 batch, four students from the 2015 batch, five students from the 2016 batch and two students from the current students' front strongly agreed that:

Government as part of the leadership of the school provides teaching and learning materials such as textbooks.

All the participants during the Focus Group Discussion agreed that:

Visual Arts Students are always taken through regular practical works by the various subject teachers.

Five students from 2014 batch, six students from the 2015 batch and seven students from the current students' front strongly agreed that:

Teachers routinely discuss marked practical works of Visual Arts Students with them.

Seven students from 2014 batch, five students from the 2015 batch and seven students from the current students' front strongly agreed that:

The school's authority provision of guidance and counselling to Visual Arts students improve their quality of work at school.

Visual Arts Students learning and academic performance is usually affected by the standard and type of the previous educational institution in which they attended that sets the parameters of students' learning outcomes. Thirty(30) participants who took part in the focus group discussions mentioned that, Dormaa Senior High School environment and certain teachers' expectations from their students also have a strong influence on students' performance.

Respondents also indicated that their performance as Visual Arts students is influenced by the nature of the previous school in which they studied; coupled with the number of facilities the school offers which usually determines its quality, which in turn affect the performance and accomplishment of the students. The kind of school attended by one influences his/her educational process in content, organization, teacher and teaching learning materials and final evaluation.

4.2.2 Teachers' Factors

The responses gathered from the field indicated that teachers contribute to certain factors that influence academic performance of Visual Arts students in Dormaa Senior High School. These factors include teachers being punctual and regular at school, their

ability to complete the syllabus, assign regular homework (practical works) to Visual Arts students all contribute to academic performance.

Comments that emerged from the respondents include the following:

Three students from 2014 batch, five students from the 2015, four students from 2016 batch and two students from the current students' front strongly agreed that:

The effective use of contact hours by teachers contributes to Visual Arts students' academic performance.

Four students from 2014 batch, seven students from the 2015 batch, three students from the 2016 batch and seven students from the current students' front strongly noted that:

Certain teachers' expectations from their students also have a strong influence on students' performance.

All the participants noted that:

The ability of Visual Arts teachers to complete syllabus is key in getting Visual Arts Students to perform creditably in the WASSCE.

The responses from the teachers also indicated that performance of teachers as reflected by the level of training and longer years of teaching experience will determine the quality of grades to be attained by students who undertake the WASSCE. A trained teacher with long years of experience seeks to possess pedagogical skills which is essential if it comes to promoting students' understanding, motivation to learn, and in so doing contributing to academic performance positively. It was revealed in the unstructured or one-on-one interview with teachers that all the four teachers in the

department have been in the service ranging from eighteen(18) years to eight years and are always alert, looking for methods and instructional materials that will promote meaningful learning.

4.2.3: Student's Characteristics

Previous Junior High School achievement affects the future academic performance of the Visual Arts Students in the school. This was confirmed by the teachers who were interviewed. They indicated that grades which are always obtained by Visual Arts students in Basic Education Certificate Examination (BECE) are also used for admission selection and which determine the students' future performance. For the students to excel academically, they have to be regular and punctual at school, attend to their assignments, initiate and embark on their own private studies, attend extra classes in school or during vacation, having access to required textbooks, etc. All these contribute to their good academic performance.

In addition to contributing to the academic performance of students in the Dormaa Senior High School, peer assistance positively influences the performance of Visual Arts students. Through peer interaction, students might increase their skills in solving both qualitative and quantitative problems.

Comments that emerged from the respondents include the following:

Three students from 2014 batch, five students from the 2015 batch, four students from 2016 batch and two students from the current students' front strongly agreed that:

Students who are self-motivated and initiate their own private studies always end up doing well academically.

All the participants agreed that:

Good grades (as in from five-ones upwards) obtained in BECE as an entry qualification of students into the Visual Arts programme contribute positively to Visual Arts students academic performance.

Six students from 2014 batch, eight students from the 2015 batch, all students from 2016 batch and seven students from the current students' front strongly agreed that:

Previous schooling and the kind of foundation obtained by students contribute to their academic performance.

All participants agreed that:

Peer teaching or assistance seeks to encourage other students' participation in class.

4.2. 4: Parent's/ Guardian Socio-economic Conditions

It needs to be established that parents or guardians' involvement in the Visual Arts students' education increases the rate of their academic performance. It is instructive to note that educated parents are able to provide further assistance to their children (Visual Arts students) regarding some of the subjects they are taught in school. During the interviews with the teachers, it was established that students who are appropriately guided by their parents or guardians do well in school.

That is, these parents or guardians show much interest in finding out about their wards' performance in school. The extent at which parents are involved in the education of the Visual Arts Students contributes to his or her performance in school. The encouragement from parents or guardians strongly improve Visual Arts students' performance in school. It was gathered that students with parents or guardians considered economically disadvantaged, suffer on the Visual Arts programme, hence end up not doing well academically at times. This is because they face challenges in getting money to buy educational tools and materials.

Comments that emerged from the respondents include the following:

Six students from 2014 batch, eight students from the 2015, all students from 2016 batch and seven students from the current students' front strongly agreed that:

The level of parent's education background can contribute to Visual Arts students' performance.

Seven students from 2014 batch, five students from the 2015, six students from 2016 batch and five students from the current students' front strongly agreed that:

Parent's socio-economic standing, which deals with factors such as their level of academic and professional qualification, income and occupational affiliation, all affect the performance of the Visual Arts Students.

4.2. 5: Home Conditions or Parental Support

Home conditions or parental support constitutes variables that can contribute to Visual Arts Students academic performance. The state of the home the Visual Arts student comes from may affect him or her since the parents are the first socializing agents in an

individual's life. Home variables such as number of siblings, survival status of parents, high dependency ratio, inability to provide textbooks and supplementary readers, inability to afford fees for extra classes, lack of motivation from parents of students can cause low academic performance.

The home vis-à-vis family structure has a great influence on the Visual Arts students psychological, emotional, social and economic state. All the respondents agreed that the state of the home affects the student's performance in school since parents are considered as the first socializing agents in the life of these students.

Comments that emerged from the respondents include the following:

All the past students from 2014 batch, five students from the 2015 batch, six students from 2016 batch and all the students from the current students' front strongly agreed that:

A Visual Arts Students desire in learning coupled with his or her conducive home environment contributes to academic performance.

All the participants noted that:

Parents continuous interaction with teachers to know the performance of their wards contribute to their academic performance.

All the past students from 2014 batch, five students from the 2015 batch, six students from 2016 batch and all the students from the current students' front strongly agreed that:

Survival status of the parents determines their involvement in the education of the student.

4.3. Objective 2: This comprises categories of responses that underpin factors that account for low academic performance of Visual Arts Students in the school.

All the respondents contacted in the field agreed that Dormaa Senior High School had registered low academic performance in the Visual Art department for the past three years (2014-2016). The following factors account for the low academic performance of the Visual Arts Students in the school.

4.3. 1: School Environment

The following constitute how the Dormaa Senior High Senior as a school, its environment and leadership influence academic performance of Visual Arts students.

Comments that emerged from the respondents include the following:

All the participants noted that:

The resources in the school do not meet the current requirements of academic work of Visual Art education.

Owusu-Afriyie, (2009) is of the view that the choice of a particular Visual Arts subject is dependent on specialist teachers available in a school and the right facilities available, tools, equipment and relevant materials.

Six students from 2014 batch, eight students from the 2015, all students from 2016 batch and seven students from the current students' front strongly agreed that:

For the past three years (2014-2016) authorities of Dormaa Senior High School has been admitting students with poor BECE grades into the Visual Arts programme hence the low academic performance registered by the final years. It is against this backdrop that certain Core Subject teachers hide behind this shortcoming in academic attainment at junior high school, and the perceived lack of interest and the generally low academic aptitude among Visual Arts students to disregard the need to give professional help to Visual Arts students in particular, to improve their potentials by giving equal attention to all subjects.

Seven students from 2014 batch, six students from the 2015 batch, all students from 2016 batch and seven students from the current students' front strongly agreed that:

Most of the teachers working in the school often have low performance expectations from the Visual Arts Students and students getting to know that their teachers have low performance expectations from them, hence it leads to poor performance by the students.

4.3.2: Teacher Factors

These are critical teachers' factors that underpin low academic performance of Visual Arts Students in Dormaa Senior High School.

All these comments emerged from the focus group discussions.

All the past students from 2014 batch, five students from the 2015 batch, six students from 2016 batch and all the students from the current students' front strongly agreed that:

Teachers in the Visual Arts department of Dormaa Senior High School do not always complete their syllabus due to a lot of constant co-curricular activities occasioned in the school.

Six students from 2014 batch, five students from the 2015 batch, seven students from 2016 batch and all the students from the current students' front strongly agreed that:

Teachers in Dormaa Senior High School do not always complete their syllabus because the time allotted for teachers to commence the academic year for first term for all first year does not work. The explanation for this being that first years, sometimes report in the middle of the first term in October. Some parents choose to bring their wards to start the programme in the second term, hence majority of students missing the whole of the first term.

Four past students from 2014 batch, seven students from the 2015 batch, eight students from 2016 batch and all the students from the current students' front strongly agreed that:

WASSCE practical works for the final year Visual Arts Students are given to them early second term. During this period, all attentions are focused on the practicals at the expense of the theory and the core oriented subjects. Hence, teachers do not get the chance to continue with the syllabus

All the participants who took part in the Focus group discussions strongly agreed that:

Certain core subject teachers perceive and label Visual Arts Students as 'unintelligent', 'not serious' and 'difficult to teach' as compared to their peers offering other programmes in the school.

4.3.3 Student's Characteristics

Low academic performance of the Visual Arts Students in the department is characterized by the following factors. Truancy plays a crucial role. The majority of the students are athletes who only become regular and punctual when there is going to be sporting events. The number of days they have been absent themselves, teaching and learning have been taking place which goes against them.

The following comments came out of the field discussions:

Six students from 2014 batch, Eight students from the 2015, all students from 2016 batch and seven students from the current students' front strongly agreed that:

Some of the Visual Arts Students come from poor homes and cannot afford buying of tools and materials to promote the progress of learning. Comparing to the elite senior high schools in the country, the majority of the Visual Arts Students in Dormaa Senior High School were admitted with low entry grades, coupled with lack of motivation from their parents or guardians and teachers, they find it difficult to excel academically.

All the participants agreed that:

Many of the Visual Arts Students perceived the theoretical aspect of their course as difficult to learn.

The effect of this perception is that these students neglect the learning of these so-called theory aspects and focus their time and effort on their elective practicals – invariably, they make poor WASSCE grades in elective subjects and miss an entry into tertiary education.

4.3.4: Parent's/ Guardian Socio-economic Condition

It was revealed from the field work that the following constitutes parent's or guardian's socio-economic conditions that contribute to low academic performance of Visual Arts Students.

These comments were couched from the discussions on the field.

Six students from 2014 batch, Eight students from the 2015, all students from 2016 batch and seven students from the current students' front strongly agreed that:

Some parents or guardians do not monitor the progress of their wards who are Visual Arts Students in the school.

All the participants agreed that:

Truancy or long absenteeism of a majority of the Visual Arts students is not checked by their parents or guardians

4.3.5: Home Conditions or Parental Support

The prevailing conditions of the home may affect the academic performance of the Visual Arts Students since the parents or guardians are considered the first socializing agents in the lives of these students. Some of the students are raised by single parents, low income parents, etc. with a lot of children to cater for. All these factors affect better living conditions which equally affect the Visual Arts Students.

These comments were couched from the discussions on the field.

Six students from 2014 batch, eight students from the 2015, all students from 2016 batch and seven students

from the current students' front strongly agreed that:

Certain parents or guardians fail to motivate their wards towards their academic progress.

Both the current and the past students who took part in the Focus group discussions indicated that their parents found it difficult to provide them with housekeeping money, hence majority of them being day-students attend classes on empty stomachs.

Seven students from 2014 batch, six students from the 2015, all students from 2016 batch and five students from the current students' front strongly agreed that:

Some of the Visual Arts Students sometimes have to absent themselves for the purposes of looking for menial jobs elsewhere to do to earn income meant to afford their personal belongings. However, some of these parents who pay their wards' fees end up not supervising the academic work of their wards.

4.4. Objective 3: Strategies to be adopted to enhance effective academic performance of Visual Arts Students in Dormaa Senior High School.

Quite a number of strategies have been indicated by respondents on what needs to be done to enhance academic performance of Visual Arts Student in Dormaa Senior High School.

Six categories were couched out from this objective to comprise the role play by the Leadership of the School and that of teachers.

4.4. 1: The Role to be Played by the Leadership of Dormaa Senior High School in Adopting Strategies to Enhance Academic Performance of Visual Arts Students in the Visual Arts department.

The leadership of Dormaa Senior High School should encourage the teachers to give more assignments or exercises (including practical works) so as to keep the Visual Arts students busy all the time. Authorities should provide more effective resources to promote efficient teaching and learning to take place. Students with interest in the Visual Arts programme should be allowed to pursue it as against pushing it for students lacking the interest.

The following comments were taken from the responses:

Seven students from 2014 batch, six students from the 2015, all students from 2016 batch and five students from the current students' front strongly agreed that:

The school authorities should provide resources.

Provision of resources and facilities in the form of furnishing the Visual Arts workshop or studio with plenty tools and materials, current Visual Arts textbooks, Information Technology or computers etc are effective in promoting effective teaching and learning in the Visual Arts department. According to Concordia University, 2013, the school authority or leadership should have an impact on students' performance which include goals setting, curriculum management, lesson planning, monitoring, resource allocation and assessment of teachers on a regular basis to help students' learning and development.

Kwesiga (2012) is of the view that the performance of students is often influenced by studies in the school and the number of facilities available which correspond to the rank of the school. This commensurate the performance and accomplishment of its students.

Eight students from 2014 batch, five students from the 2015, all students from 2016 batch and six students from the current students' front strongly agreed that:

The school's authorities should identify all the athletes in the Visual Arts department who are facing financial difficulties and give them scholarship.

Those Visual Arts Students who are athletes should be given a scholarship for them to be motivated to learn at school and at the same time excel on the various athletic disciplines they are good at.

All the participants agreed that:

The school itself must patronize the art works produced by Visual Arts Students and use them for decorative purposes in the school.

It will provide a sense of joy, appreciation and encouragement to the Visual Arts Students if the authorities of Dormaa Senior High School partake in patronizing all the art works or artefacts to be showcased during annual exhibitions to be organized by the Visual Arts Department.

Six students from 2014 batch, five students from the 2015, all students from 2016 batch and five students from the current students' front strongly agreed that:

Enabling school's premise and atmosphere promote positive academic performance.

How a school environment is and its teachers' expectation from its students are strong factors that influence a student's performance (Considine and Zappala, 2002 as cited by Sparkles, 1999). A teacher who works in a poor school setting might run into the lack of basic facilities which is required for academic work to take place and when students perceive that teachers are expecting a low performance on their part, it is likely to result in that.

A significant factor in learning as stated by Ehiametor (1990) and Farrant (1992) is the learning environment and a conducive atmosphere is what makes learning more effective.

Conducive school's climate necessitated by improved teaching methods, provision of available teaching aids, tools for practicals, avoiding overcrowding in the classrooms

and Visual Arts workshops, etc. are considered as factors that require attention to ensure academic success among Visual Arts Students in Dormaa Senior High School.

Five students from 2014 batch, six students from the 2015, all students from 2016 batch and five students from the current students' front strongly agreed that:

The school authorities must create the Visual Arts classroom as a social context where communication occurs freely.

It needs to be pointed out clearly that with teachers and students having their roles to play, and with opportunities for collaboration structured to influence students' understanding and construction of knowledge, it thus affects learning. Class size affects academic performance of Visual Arts Students because students' performance lowers as class size increases.

Similarly, overcrowding in the Visual Arts studio makes it difficult for the teachers to manage each student's attention and also make use of various teaching and assessment methods during practicals. The fact is, where a teacher is limited by space and finds it difficult to provide individual attention and supervision, students who are unattended to tend to disturb the entire class and distract the attention of other students during lessons. The high level of distractibility as evidenced in the Visual Arts—studio, the density of class size, and social interaction that goes on with certain students are potential barriers to high academic performance.

In addition, the responses revealed that there should be constant monitoring of teaching and learning processes, attitudinal change and motivation could build the capacity of Visual Arts students to participate fully in the economic development of Ghana.

The school authorities should ensure constant and proper collaboration between teachers who teach in the department. This will seek to encourage and motivate the Visual Arts Students to study hard what is taught them so they would get better grades in WASSCE and get the opportunity to further their education to the tertiary level.

4.4. 2: Specific Roles to be Played by Teachers in Adopting Strategies to Enhance Academic Performance of Visual Arts Students

Teachers as technical handlers should vary their teaching methods and techniques when delivering instructions in both Core subjects and Visual Arts subjects to help Visual Arts Students considered as seemingly weak to better understand, encouraged to learn topics taught them and excel.

In this era, according to Bubblews (2013, p. 3) teachers should "develop learners instead of teaching them, who help their pupils to become independent (learning to learn), who provide students with motivation and interest for lifelong learning and urge them to become autonomous learners". Teachers make use of a number of techniques in order to improve academic goals. This involves evaluation, assessment and provision for students who require special attention as far as academic work is concerned.

Teachers should prepare their lesson notes in advance before going to the classroom to teach. Students who do not do well academically should either be repeated or sacked after receiving several warnings against withdrawal from the authorities.

The following comments were couched to validate the findings.

Seven students from 2014 batch, six students from the 2015, all students from 2016 batch and five students from the current students' front strongly agreed that:

Teachers must identify weak students and offer them with special assistance.

Individual differences in terms of abilities require teachers to identify weak students and offer them with special assistance. This is because Visual Arts Students vary in the way they acquire new information or skill. Hence, no single teaching method or strategy can satisfy all needs of Visual Arts Students due to the technical and vocational nature of the Visual Arts Programme.

Six students from 2014 batch, five students from the 2015, all students from 2016 batch and seven students from the current students' front strongly agreed that:

Teachers in the Visual Arts Department must identify the intelligence strengths of Visual Arts Students so that they can teach to reinforce the strengths of their intelligence and learning styles when teaching new topics.

For this strategy to materialize and achieved in Dormaa Senior High School, it is expected of the teachers who teach both the Core subjects and the Visual Arts subjects to vary their teaching methods and techniques when delivering instructions for the purposes of helping those students who are considered seemingly weak to improve upon their lot. This is because teaching strategies and methods have great influence on the entire academic performance of Visual Arts Students. Effective teaching methods can combine both the theory of multiple intelligences and the concept of learning styles, as Visual Arts Students vary in the way they acquire new information or skills when it

comes to executing their practical works and learning the theoretical aspects of their education.

All participants agreed that:

Some of the teachers must be talked to, to change their attitudes towards their job as teachers.

This is sometimes reflected in the teachers' poor attendance to class, lateness to school, passing of unpleasant comments about Visual Arts Students academic performance that could damage their ego, confidence and interest towards learning.

All the participants agreed that:

Heads of Department must consider an area of specialization of teachers before allocating subjects to them.

Quality of Visual Arts teachers and their commitment to the teaching job are key inputs that can contribute to the positive academic performance of the Visual Arts Students. In essence, teachers' knowledge of the subject matter coupled with the choice of the right textbooks, instructional time and other learning materials has great and positive influence on the academic performance of the Visual Arts Students during WASSCE. In other words, teachers in the department without both the academic and the professional qualification would undoubtedly contribute to a negative influence on the teaching and learning of any of the Visual Arts subjects entrusted into his or her hand. It was revealed

during the interview with the teachers that out of the four teachers in the department, two are professional teachers and two are not.

The academic performance of the Visual Arts students is positively correlated with teachers' qualification. Thus the teachers' qualification is considered to be an effective causal effect on Visual Arts students' performance

All the participants agreed that:

For the Visual Arts students to achieve academic performance, language and communication styles can be blended during teaching and learning processes.

Academic performance of Visual Arts students has been always dependent on the type of language used in school which is familiar to the students. The majority of the participants opined that the Visual Arts Programme being technical and practically oriented, sometimes should be taught in the language the Visual Arts Students is also familiar with so that as learners, they can easily comprehend every aspect of the instructional content. The high failure rates experienced by the Visual Arts students in Dormaa Senior High School is as a result of the English Language used in delivering the instructions in school which differs from the various local dialects spoken at home. This means that language and communication styles if not checked in the classrooms can therefore impede the academic performance of the Visual Arts students.

All the participants agreed that:

Teachers with long years of teaching experience have significant effect on the academic performance of Visual Arts Students.

Effective teaching of the Visual Arts subjects by experienced teachers who have taught for at least ten years can result to their mastery of the subject. Experienced teachers can boast of effective communication in the classroom, effective lesson preparation, planning and presentation, distribution of tests by taking into account individual differences, and allowing Visual Arts Students to practice and apply theories taught them. Also, experienced teachers can create conducive learning environments that are safe and supportive of their students to ensure high teaching and learning outcomes. Such teachers are best when it comes to establishing learning goals for their students and cultivate a positive culture that enables students to perform positively in their academics. Thus, experienced teachers know that no single teaching strategy or method can satisfy all needs of Visual Arts Students hence varying the methodology can be effective. Based on the interview conducted, it was found that the teachers in the department have been in the teaching profession, ranging from 18 years, 12 years, 8 years and 4 years.

As learning deals largely with student's ability to organize and use ideas and skills to solve a problem, Visual Arts Teachers ought to deliver instructional content that their students need to know, understand, and be able to practicalise. However, the whole teaching exercise is rendered unproductive if students are not actively involved in the learning processes and experiences.

4.4.3: Specific Roles to be Played by Visual Arts Students' themselves in Adopting Strategies to Enhance their Academic Performance.

During the Focus group discussions, all the participants agreed to the fact that Visual Arts students' themselves have roles to play when it comes to contributing to strategies to enhance their academic performance.

The following responses emerged in the various discussions held between the researcher and the participants.

All the participants agreed that:

Visual Arts Students' should be willing and able to learn what is taught them by their teachers in class.

Willingness stems from the extent at which the Visual Arts Students' will be disciplined towards their studies as senior high school students. Also, discipline on the part of the Visual Arts Students' include they developing inordinate passion towards learning both what they have been taught and new things they have not in class. For example, making friends with Visual Arts students from the elite schools for the purposes of comparing notes, hence having an idea about what they have been taught and the ones they have not. It is also believed that those elite schools always lead the less endowed senior high schools in infrastructure, facilities, experienced teachers, teaching and learning aids, etc. Developing the culture of executing practical works assigned to them by their teachers and solving past questions on their own can also be included in the "passion package".

It is a rule of thumb that per the WAEC rules, failure on the part of any Visual Arts Students to execute his or her practicals works amount to overall failure in the entire WASSCE. Visual Arts Students willingness and ability to learn automatically prepares their minds and souls to receive and assimilate whatever is taught them by their teachers.

Though students with lower admission requirements are sometimes admitted, with the willingness on their part they can take their studies seriously and excel at all cost.

Visual Arts students should avoid attaching themselves to other deviant students (e.g. truants, those who do not do their practicals, etc.) but the serious and purposeful ones.

All the participants agreed that:

Visual Arts students' backgrounds influence their academic performance.

Visual Arts students from different backgrounds and cultures always attend senior high with differing prior knowledge and resources to learning. The background of a Visual Arts Students affects positively the teaching-learning process and also serves as a springboard for further education. To succeed in identifying and helping students who are not learning requires teachers to ensure that instructions being delivered are not unsuitable to the students' learning styles. It is therefore imperative that academic performance of Visual Arts Students is investigated so that the negative factors that affect students' performance could be lessened to enable all students to have equal opportunity to excel on the Visual Arts Programme.

Some of them complain that sometimes they are faced with financial difficulties. To handle this issue, Visual Arts students are advised to be innovative by producing and packaging their practical works and sell them to interested buyers for handsome fees.

Such proceeds obtained from such sales can be used by the Visual Arts Students to help them solve their financial needs.

All the participants agree that;

Visual Arts Students should be regular, punctual in all classes and avoid absenteeism.

Those students who have cultivated the habit of not attending classes should desist from such behaviour. Absenting themselves from classes are catalyst for them to missing several topics being taught behind them. The Visual Arts Students must be counseled to be punctual and regular in classes, so that they will get the opportunity to participate in class activities such as executing their practical works, taking part in the theory sections, etc.

Seven students from 2014 batch, eight students from the 2015, all students from 2016 batch and seven students from the current students' front strongly agreed that:

They should use their time judiciously.

It was revealed that the majority of the Visual Arts students spend much time on social media. The Visual Arts students should be counseled for them to use those precious times on social media to learn all the topics they have been taught by their teachers. Even the data they have been using to browse for social media can be used to do further research concerning course contents.

4.4. 4: Specific Roles to be Played by Parents and Guardians in Adopting Strategies to Enhance the Academic Performance of Visual Arts Students.

All students from 2014 batch, eight students from the 2015, all students from 2016 batch and seven students from the current students' front strongly agreed that:

Parents and guardians irrespective of their socio-economic condition should do their possible best in providing for the Visual Arts Students.

Majority of Visual Arts Students who took part in the Focus group discussions did indicate that they perform bad academically because they were sometimes faced with financial difficulties in accessing reading materials, tools for practicals. They blame it on their guardians or parents' inability to providing for their wards in terms of buying books, tools for practical, money to buy food, and other personal belonging. The Visual Arts Students lacking of all these essential things do not contribute positively to academic performance.

All the participants agreed that:

Parents and Guardians should take upon themselves to advise their wards who are Visual Arts students about the essence of education so that they will be encouraged to study the programme seriously.

After providing for their wards, the next appropriate thing is for the parents or guardians to regularly advise their wards to take their studies seriously.

Eight students from 2014 batch, six students from the 2015, all students from 2016 batch and seven students from the current students' front strongly agreed that:

Parents and guardians should encourage and motivate the Visual Arts Students with rewards.

In their quests of getting their wards doing well in school, parents or guardians must promise them of rewards so that the students can be motivated to take their studies in school seriously.

All the participants to the Focus group discussions agreed that:

Parents or guardians who are educated should find time to supervise and monitor the homework of their wards who are doing Visual Arts.

After providing for the students, it should not end there. Parents or guardians must take it upon themselves to sometimes pay their ward unannounced visits in the school to find out from their teachers the academic performance of their wards. Where the children fall short, parents can sit with their wards, discuss these shortfalls with them and provide solutions to help contribute positively to their academic performance.

Parents and guardians regardless of their work schedule should find time to visit the school to find out about the academic performance of their wards.

4.4. 5: Specific Roles to be Played by the Dormaa Ahenkro Community in Adopting Strategies to Enhance the Academic Performance of Visual Arts Students

There was overall agreement regarding the role the entire community of Dormaa Ahenkro needs to play to enhance the academic performance of the Visual Arts Students in Dormaa Senior High School.

Five students from 2014 batch, Four students from the 2015, six students from 2016 batch and seven students from the current students' front strongly agreed that:

There should be the institution of the Community Watch committee to be emerged from the Unit Committee to put close surveillance on Visual Art Students who are truants, and other students who always hide in obscure places during class hours and bring them to the school. The school authority, and especially the Visual Arts teachers should then monitor the attendance of these students.

The existence of the Community Watch committee will lead to the curbing of the regular Visual Arts Students absenteeism and truancy.

Six students from 2014 batch, Six students from the 2015, all students from 2016 batch and seven students from the current students' front strongly agreed that:

In the community, people with houses should help by renting them out to Visual Arts Students who are day students at affordable fees.

Most of the students who do not get access to the boarding house complain about getting access to accommodation outside the school's premises.

All the participants agreed that:

The school's authorities must give permit to food vendors who can prepare healthy food to sell to students on time and on the school's premises, so that Visual Arts Students will not use it as excuse to run away to town. Apart from this, the food sold should also be affordable.

Most of the Visual Arts Students who go out of the school's premises during break time and do not come back, do so because enough food vendors are not in the school's premises for them to buy from them. The few who sell to them on the school's premises to buy, according to the students sell at exorbitant price. Should the permission be given and enough food vendors are seen around the school's premises the habit of running away during break time will minimize.

All the participants agreed that:

Visual Arts Department should organize regular exhibitions to showcase the various innovative art works of Visual Arts Students; with invitations given to the community to participate in the event.

From first to the third year, all the Visual Arts Students must be encouraged to produce several artworks or artefacts for the purposes of promoting and marketing them to the general public. The sales that will be made from selling those artefacts can be used by the Visual Arts Students to support themselves while in school. A way of solving their financial challenges

All the participants agreed that:

Job owners in the community should accept employing Visual Arts Students as casual workers during the long vacations to enable them access funds to buy their personal belongings.

Some people in the community who own businesses like poultry farms, shops, filling stations, etc. should give opportunity to the Visual Arts Students to work with them as a way of raising income to buy their personal needs to help them perform well academically.

All the participants agreed that:

Landlords and landladies accommodating the Visual Arts Students can play the role of guardians by advising them to take their studies seriously, especially, the Visual Arts Students who have rented accommodations but do not attend classes. These landlords or land ladies can report them to the school authorities or the Community Watch committee to be organized to monitor truancy.

4.4. 6: Specific Roles to be Played by Old Students in Adopting Strategies to Enhance the Academic Performance of Visual Arts Students

Old students here refer to past students of the Visual Arts Department of Dormaa Senior High School.

Six students from the 2014 batch, Six students from the 2015 batch, all students from 2016 batch and seven students from the current students' from strongly agreed that:

Invitations should be given to these old students to attend the annual Visual Arts

Department exhibitions to be organized by the Visual Arts Department.

During the exhibitions, the invitees will use the occasion to address the Visual Arts Students on topics like the essence of taking their studies seriously, avoiding bad companies, shunning drugs, etc. These invitees should be personalities who were former Visual Arts Students and now serving mother Ghana in various capacities. This will encourage the current Visual Arts Students in Dormaa Senior High School that, should they do well in school and pass their exams successfully, there will be several job opportunities out there for them.

All the students from 2014 batch, all the students from the 2015 batch, Seven students from 2016 batch and seven students from the current students' front strongly agreed that:

Awards should be instituted by any of the old student groupings and be presented to the best and deserving Visual Arts Students.

This will encourage others who are not doing well academically to emulate the brilliant ones.

Seven students from 2014 batch, eight students from the 2015 batch, seven students from 2016 batch and seven students from the current students' front strongly agreed that:

Any of the old students grouping should be written to, to come and help the Visual Arts

Department in terms of expanding the Visual Arts workshop and resourcing it regularly.

Resourcing the Visual Arts workshop will help the students to access the modern tools and logistics which help the Visual Arts Students to excel in their WASSCE.

All the participants agreed that:

Old students can liaise with the business community to provide various forms of scholarship incentives to those needy, but brilliant Visual Arts Students—hence lessen their financial burden.

Six students from 2014 batch, six students from the 2015 batch, all students from 2016 batch and seven students from the current students' front strongly agreed that:

The Visual Arts Department should take advantage of the annual Speech and Prize Giving Ceremony to invite old Visual Arts Students who are serving mother Ghana and the world at large in various capacities to counsel the students and also donate to the department either in cash or in kind.

The guidance and counselling that will be given to the Visual Arts Students by these old students will go a long way in changing those students who are not serious to turn new leaves.

Five students from 2014 batch, five students from the 2015 batch, all students from 2016 batch and seven students from the current students' from strongly agreed that:

The old students in collaboration with the media, the Churches and the mosque should take it upon themselves to promote up and coming exhibitions of the Visual Arts Department..

This will bring more people from all walks of life to come and participate in the exhibition to be organized by the Visual Arts Department. Patronage of the artworks will go up during such exhibitions and students who are the producers of such products will get money from the sales to be made to support themselves while in school.

AND SANE

CHAPTER FIVE

SUMMARY, CONCLUSION AND RECOMMENDATIONS

5.1. Overview

This chapter presents summary of all the study's findings, and their conclusions. What follows next are the recommendations for policy formulation and suggestions given to areas of further research.

5.2 Summary

5.2.1 Summary of Findings

The findings revealed, among other things that:

It is indicated in chapter four that there are certain prevailing factors that influence academic performance of the Visual Arts Students in Dormaa Senior High School.

These include recruiting of teachers with high academic qualification, provision of learning tools and materials as resources, judicious use of contact hours by teachers, quality of medium of instructions adopted in the school, tuition trends, daily study by Visual Arts Students, provision of guidance and counselling to parents or guardians for creating a positive home environment for improvement in Visual Arts Students' education. The direction from the school authorities sometimes influences performance of their students.

In specifics, the school's leadership should supervise academic progress of the various departments of the school; organises workshops for the students to help improve their academic performance; ensure there is discipline to promote a peaceful environment for

teaching and learning to thrive; evaluate or appraise the performance of teachers at the end of every academic year. On the perspective of teachers of Dormaa Senior High School, they influence academic performance of Visual Arts Students in the school through being punctual and regular at school, ability to complete one's syllabus, assign regular assignments or practical works to Visual Arts Students to keep them busy. On the Student's characteristics or demographics, the following characteristics affect Visual Arts Students academic performance.

Previous Junior High School attainment affects the future academic performance of the Visual Arts Students in the school. This was confirmed that the grades obtained by a Visual Arts Students in BECE that are used for admission selection determines a student's future performance.

Others include the Visual Arts Students being regular and punctual at school, attend to assignments, embark on private studies, attend extra classes in school or during vacation, having access to required textbooks, etc. all contribute to their academic performance.

On the side of parents or guardians of Visual Arts Students, their involvement in their wards' education increases the rate of the student's academic performance. Educated parents are able to provide further assistance to their children regarding some of the subjects.

All the respondents contacted in the field agreed that academic performance of the Visual Arts Students in Dormaa Senior High School for the last three years (2014-2016) have been low.

The following factors account for the low academic performance of the Visual Arts Students in Dormaa Senior High School: Lack of resources in the Visual Arts Department which do not improve academic work of visual Art education, admission of students with poor BECE grades into the Visual Arts department, and perceived lack of interest of Visual Arts Students regarding the programme. There is also the inability of teachers in the Visual Arts Department to complete their syllabus.

On the part of the Student's characteristics as factors that underpin low academic performance of Visual Arts Students in Dormaa Senior High School, the following can be summed up.

Visual Arts Students sometimes focus heavily on the elective practicals to the neglect of learning the theory aspect which results in them making poor WASSCE grades in elective subjects and eventually miss the entry into tertiary education.

The following are given as strategies to be adopted to enhance effective academic performance of Visual Arts Students in Domaa Senior High School.

The leadership of the school should encourage teachers to give a lot of assignments and exercises (including practical works) so as to get the Visual Arts Students busy all the time. Authorities should provide more effective resources to promote efficient teaching and learning to take place in the Visual Arts department.

Students with the interest should be admitted to pursue the Visual Arts programme as against pushing it for students with low grade and no interest.

Teachers should endeavour to vary their teaching methods and techniques when delivering instruction in both Core subjects and Visual Arts oriented subjects to help Visual Arts Students who are considered as seemingly weak to better understand, encourage to learn topics taught and excel.

Also, the majority of the Visual Arts students who are athletes, should be given scholarship by the member of parliament(MP) of the contituency and the chief of the traditional area to motivate them to pursue the programme and to excel. The school authorities should ensure constant and proper collaboration between teachers of both Core Subjects and Visual Arts to counsel the students on how to plan, allocate, manage and use their time judiciously on all the aspects and subjects making up of the Senior High School curriculum.

During exhibitions to be organized annually by the Visual Arts Department, the authorities of the school must patronize the art works produced by Visual Arts Students and use them for decorative purposes in the school, provides an enabling atmosphere to promote positive academic performance, and also create the Visual Arts classroom as a social context where communication can occur freely.

Teachers need to perform the following roles in order to get Visual Arts Students to perform positively academically.

They must identify the strengths of students so that they can teach to reinforce the strengths when teaching new topics.

Also, some of the teachers must be talked to by management to change their attitudes towards their job as teachers.

Heads of Department must consider an area of specialization of teachers before allocating subjects to them; Language and communication styles can be blended during teaching and learning processes; and finally teachers with long years of teaching experience have significant effect on the academic performance of Visual Arts Students.

The Visual Arts Students themselves must adopt the following strategies to help them perform well academically:

Visual Arts Students should be encouraged and motivated to learn what is taught them by their teachers in class.

Visual Arts Students' backgrounds influence their academic performance; Visual Arts Students should be regular, punctual in all classes and avoid absenteeism; they should be counseled for them to use those precious times on social media to learn all the topics they have been taught by their teachers.

5.3 Parents and guardians must adopt the Following strategies to enhance the academic performance of Visual Arts Students:

Parents and guardians irrespective of their socio-economic conditions should do their possible best to provide for their wards. They should take upon themselves to advise the Visual Arts Students on the essence of education so that they will be encouraged to study the program seriously. They should encourage and motivate the Visual Arts Students with rewards. Parents or guardians who are educated should find time to supervise and monitor the home works of their wards who are Visual Arts Students.

5.4 The Dormaa Ahenkro Community should adopt the following strategies to enhance the academic performance of Visual Arts Students

The community should appoint some people to form a committee known as Community Watch committee to put close surveillance on Visual Arts Students who are truants, always hiding in obscure places during classes hours and bring them to the school; people with houses should help by renting them out to Visual Arts Students who are day students at low prices. Thus, instead of paying one year advance, the students should be given the opportunity to pay in monthly instalments.

Studies have shown that when parents participate in their ward's education, it has a positive effect on their academic performance and the school's success (Halsey, 2004; Christie, 2005). Involvement of parents in their ward's education improves the morale of teachers and thereby may create a cordial relationship between the school and the community. The important aspect of it is that of parents involving themselves in their ward's education. There are several things that parents or guardians can do to help their ward's education. The purpose of the involvement promotes positive behaviour between students, checking up on students' work, encouraging contributions from each parent towards their ward's education, helping students to do their assignments, tutoring, supervision of afternoon study periods, coaching sports and motivational talks.

The school authorities must give permission to food vendors who can prepare healthy foods and sold to students on time and on the school's premises, so that Visual Arts Students will not use it as an excuse to run away to town. Visual Arts Department should organize regular exhibitions to showcase the various innovative art works of Visual Arts Students; with invitations given to the community to participate in the event;

Job owners in the community should accept employing Visual Arts Students as casual workers during the long vacations to enable them access funds to buy their personal belongings.

Finally, landlords and landladies accommodating Visual Arts Students who are not their wards can play the role of guardians by advising them to take their studies seriously.

5.5 Old students need to adopt the following strategies to enhance the academic performance of Visual Arts Students

These old students need to attend the annual Visual Arts Department exhibitions to be organized by the Visual Arts Department.

Awards should be instituted by any of the old students and be presented to the best and deserving Visual Arts Student. The old students from the visual arts department should be written to, to come and help the Visual Arts Department in terms of expanding the Visual Arts workshop or studio and resourcing it regularly. Old students who are businessmen in and around Dormaa community can provide various forms of scholarship incentives to those needy, but brilliant Visual Arts Students, hence, lessen their financial burden. The Visual Arts Department should take advantage of the annual Speech and Prize Giving Ceremony to invite old Visual Arts Students who are serving mother Ghana and the world at large in various capacities to address the students.

Lastly, the old students can invite the media, the Churches and the mosques to take it upon themselves to promote up and coming exhibitions of the Visual Arts Department.

5.6 Conclusions

It needs to be noted with emphasis that like any academic institution, where there is greater focus on the academic attainment of students, who are considered as the core business of the school, all the factors that affect students' academic performance in the Visual Arts department of Dormaa Senior High School need to be identified and improved upon. This is to say that the findings of this study have confirmed with extant knowledge which posit that there are certain number of factors that affect academic performance of students in general. That gives much to conclude that the findings of this study are congruent with the existing findings (e.g. Emerson and Goddard, 1993; Witziers, Bosker and Kruger, 2003; Strauss, 2013; Wallace, 2011; Afe, 2001; Taal, 1996; Kwesiga, 2002, Crosnoe and Elder, 2004; Harb and El-Shaarawi, 2006; Reid, 1995; Pashler, McDaniel, Rohrer, and Bjork, 2008; Sentamu, 2003; Crosnoe and Elder, 2004).

The study's findings have identified that the Visual Arts Students in Dormaa Senior High School for the past three years, that is, (2014-216) have recorded low academic performance for the students in the WASSCE. There are specific factors that underpin the low academic performance recorded by Visual Arts Students in Dormaa Senior High School Visual Arts Department among others include: majority of the Visual Arts Students enter with poor BECE grades, teachers inability to complete the syllabus, final

year Visual Arts students heavily focus on the elective practical aspects to the neglect of the theory aspects which always results in majority of them missing out the opportunity to continue their education up to the tertiary level.

Lastly, it is concluded that the school authorities and other relevant stakeholders have a significant role to play in dealing with the factors that underpin low academic performance of Visual Arts Students in Dormaa Senior High School. Strategically, leadership of Dormaa Senior High School should encourage teachers to give more assignments or exercise (including practical works) so as to get the Visual Arts Students busy all the time; and to provide more effective resources to promote efficient teaching and learning to take place in the Visual Arts department. The authorities should ensure constant and proper collaboration between teachers of both Core Subjects and Visual Arts subjects. Also to counsel the students on how to plan, allocate, manage and use their time judiciously with all the subjects making up the Senior High School curriculum. All these established as the findings indicate consistency in existing studies (e.g. Gagne, 1990, Ehiametator; Farrant, 1982; Moore, 1998; Mortimore et al., 1998; OFSTED, 1995 Zapalska et al, 2002; Mordedzi, 1999; Witkin and Berry, 1975; Witkin and Goodenough, 1981; Dunn et al., 1990; 1989; Doyle, 2006, as cited in Mucherah and Frazier, 2013; Babatunde and Olanrewaju, 2014).

5.7 Recommendations for Policy Formulation

Although, most of the questions were answered positively by respondents on the field, findings have indicated that there is more to be done as far as adopting strategies to enhance academic performance of Visual Arts students is concerned. The researcher therefore recommends the following:

The poor academic performance of students in Visual Arts in WASSCE should be treated as the topmost agenda at the Visual Arts Department, Parents/Teachers Association, Old students' meetings. In such meetings, ideas and strategies needed to solve the problem will be brainstormed by the attendees, which will be adopted by the authorities to help them in solving that challenge bedeviling the school in recent times.

The authorities of the Dormaa Senior High School should continue their quests in recruiting a number of teachers with high academic qualification and experience. They will have the required expertise in handling all caliber of students who may be admitted into the school.

The school authorities should also make judicious use of Information Technology systems. By this, they will have most of their projects, activities and processes (in terms of monitoring and supervision) speed up. Hence, efficiency and effectiveness will be achieved all the time.

To gain more experience, teachers of Visual Arts must register with WAEC as examiners for them to abreast themselves with how questions are set; where questions are set from the syllabus and what specific questions sometimes are needed to be given to students as exercises, assignments. The more they contribute their quota in helping

the examining body in carrying out its mandate as the test administrator (specifically, appointed as examiner or markers), Dormaa Senior High School will benefit from their expertise gained through knowledge transfer. Teachers must continually motivate all Visual Arts Students in Dormaa Senior High School. They should use their humble beginnings to counsel the students in improving their lots. The supervision of teachers' works at the school should be intensified so that the effective use of contact hours by teachers can contribute to the effective completion of the syllabus hence contributing to the overall academic performance of the Visual Arts Students.

The guidance and counseling unit should be structured by the school in a way so as to appeal effectively to the Visual Arts Students. This will help solve those consistent cases of truancy, no urgency with regards to initiating their private studies, attending classes, peer assistance, having access to reading materials- textbooks, past questions, etc. It was revealed in the study that the active participation of most of the Visual Arts Students in athletics or sporting disciplines contributes to their low performance in the Visual Arts Programme. To realize the all-round development of the Visual Arts Students, curricula should be combined with co-curricular (extra-curricular activities). Co-curricular activities help in the realization of the aims and objectives of education. Games and Sports make them mentally and physically fit and sound. Games make them learn how to perform while losing or wining an event. Sometimes, classroom teaching becomes monotonous and routine. Here, co-curricular activities can bring pleasant and joyous experiences. Co-curricular activities not only make the students active and energetic, but also enable them to harness the in-depth potential of students. It was reported that WASSCE practical works for the final year Visual Arts Students are always given to

them early second term, hence in order to speed up and get it executed to meet the submission deadline, this always compel the Visual Arts Sstudents to focus all their attention towards the practicals at the expense of the theory and the core oriented subjects. This makes the teachers face the challenge in continuing with the syllabus and get it completed. In dealing with this challenge, it is recommended that the authorities of Dormaa Senior High School should appeal to WAEC as the test administrator to either bring the practicals questions earlier than the time the schools have been receiving them or they extend the submission deadline to an additional two or three months. With this, the Visual Arts Students can apportion their time equally on both practical works, theory and the core subjects. If this appeal from the school fails, the authorities can go ahead and change the strategy by using agencies like the Ministry of Education, CHASS, GNAT and NAGRAT as a way of putting pressure on WAEC to heed this positive call.

Teachers of core subjects in Dormaa Senior High School as a matter of urgency must be advised to change the way they address the Visual Arts Students in the classroom. This will let them stop the use of certain derogative remarks (such as 'unintelligent', 'not serious' and 'difficult to teach' as compared to their peers offering other programmes in the school) that they have been using for the Visual Arts Students which to some of the students they are suppressed anytime such words are uttered by the teachers to describe them.

Lastly, old students and other stakeholders should help the school by providing effective resources(that is, furnish the Visual Arts workshop or studio with plenty tools and materials, current Visual Arts textbooks, Information Technology computers, etc.) to promote effective teaching and learning in the Visual Arts department. There should be

constant monitoring of teaching and learning processes, attitudinal change and motivation to build the capacity of Visual Arts students to participate fully in the economic development of Ghana. The school authorities should ensure constant and proper collaboration between teachers who teach in the department. This will seek to encourage and motivate the Visual Arts Students to study what is taught them so they would get better grades in WASSCE and get the opportunity to further their education to the tertiary level.

5.8 Suggestions for Further Studies

It is suggested, among other things that future researchers should turn to investigate into the following:

- 1. Improving Visual Arts Education in Dormaa Senior High School. Here the future study will delve into how the aspects of Visual Arts handled in the school can be improved. Objectives of the study will trace the emergence of Visual Arts in the school, focus on the role of stakeholders, strategic interventions to be adopted for improvement, possible challenges, etc. The main finding of this current study, "factors affecting academic performance of Visual Arts" indicate that Visual Arts Students in Dormaa Senior High School record low academic performance in recent times. So this topic will bring to the fore and add to the existing ones strategies authorities in schools in Ghana facing similar challenges.
- 2. The role of effective medium of instructions in improving Visual Arts Education in Ghana. A medium of instruction is a language used in teaching. It may or may

not be the official language of the country or territory. If the first language of students is different from the official language, it may be used as the medium of instruction for part or all of schooling. So in examining the role of effective medium of instructions in improving Visual Arts Education, future researchers can take the bold attempt at focusing on the actual language(s) needed to be used in teaching Visual Arts Students, the benefits to be derived in adopting such languages, possible challenges to be dealt with.

3. The roles played by co-curricular activities in affecting academic performance of Visual Arts Students in Ghana. Activities sponsored or recognized by a school or college which are not part of the academic curriculum but are acknowledged to be an essential part of the life of an educational institution. Co-curricular activities include sports, school bands, cadet, red cross, etc. In the current study, "factors affecting academic performance of Visual Arts" it was revealed that co-curricular activities such as sports contribute to the low academic performance of Visual Arts Students in the school. It is important for future researchers to investigate how the concept affect academic performance of Visual Arts Students positively and negatively, how they can be streamlined to better their purposes in the schools.

REFERENCES

- "What is HRBAP? Human Rights-based Approach to Programming. UNICEF".

 UNICEF. Retrieved 2016-09-28. Retrieved January 15 2018 from https://www.unicef.org/policyanalysis/rights/index_62012.html
- Afe, J. O. 2001. Reflection on becoming a teacher and the challenges of teacher education: Inaugural lecture series 64. Benin City: University of Benin.
- Ankomah, A. Y. 2002. The success story of private basic schools in Ghana: The case of three schools in Cape Coast. Journal of Educational Management, 4, 1-4.
- Ali,S., Zubair Haider, Z., Munir, F., Khan, H.,Ahmed-, A. 2013. Factors Contributing to the Students' Academic Performance: A Case Study of Islamia University Sub-Campus. *American Journal of Educational Research*. 2013, 1(8), 283-289. DOI: 10.12691/education-1-8-3. Published online: August 25, 2017
- Amedahe, F. K. 2002. Notes on Educational Research. Unpublished lecture notes.

 University of Cape Coast
- Amedele, 2004. Research Design and methods of Data collection and analysis: Researching students Conceptions in a multiple-method case study.
- Asihene, G. 2009. The role of core subject teachers in the academic performance of Visual Arts students in Ghanaian Senior High School. Master's Thesis. Kwame Nkrumah University of Science and Technology, Kumasi, Ghana.
- Babatunde, M. M. and Olanrewaju, M. K. 2014. Class size and school climate ad correlates of secondary school student's scholastic achievement in Itesiwaju Local Government Area of Oyo State, Nigeria. Global Journal of Instructor of Human Social Science: & Linguistic & Education, 14 (3) 14-21.
- Babbie, E., 2011. The Basics of Social Research, (5th Edition). Wadsworth Cengage Learning.
- Bandura, A., 1977. Self-efficacy: Toward a unifying theory of behavioral change. Psychological Review, 84, pp.191–215.
- Bandura, A., 1986. Social foundations of thought and action: A social cognitive theory. Englewood Cliffs, NJ: Prentice-hall.

- Bandura, A., 1997. Self-efficacy: The exercise of control. New York: W.H. Freeman and Co.
- Basit, T. N., 2010. Conducting research in educational contexts. London: Continuum
- Battle, J. and Lewis, M., 2002. The increasing significance of class: The relative effects of race and socioeconomic status on academic achievement. *Journal of Poverty*, 6(2), pp.21-35.
- Baverstam, O. and Larsson, M., 2009. Strategic Green Marketing: A Comparative Study of How Green Marketing Affects Corporate Within Business to Business: Bachelor of Marketing Thesis: Lulea: Lulea University of Technology.
- Borde, S. F. (1998). "Predictors of Student Academic Performance in the Introductory Marketing Course". Journal of Education for Business, 73 (5), 302 307.
- Bell, J. M. (2012). Define academic performance. eHow Contributor. Available from www.google.com
- Braun, Virginia and. Clarke, Victoria (2006) Using thematic analysis in psychology. Qualitative Re- search in Psychology, 3 (2). pp. 77
- Bray, M. (2007). The shadow education system: Private tutoring and its implications for planners. Paris: IIEP, UNESCO Available online at:

 http://unesdoc.unesco.org/images/0011/001184/118486 e.pdf
- Bresó, E., Schaufeli, W.B. and Salanova, M., 2011. Can a Self-efficacy-based Intervention decrease Burnout, increase Engagement, and enhance Performance? A Quasi experimental Study. *Higher Education*, 61, pp.339–355.
- Brunard, P. 1991. A method of analysing interview transcripts in qualitative research.

 Nurse Educ Today, 11(6), 461-6.
- Bryman, A.and Bell, E. 2007. Planning a research project and formulating research questions. In: Business Research Methods. New York. Oxford University Press. P. 75-92.
- Bryman, A. and Bell, .(2011. Business Research Methods 3e OUP Oxford, University Press

- Bubblews, 2013. The role of teachers in the 21st century education. Retrieved: 17 May 2013 from http://www.bubblews.com/news/279989-the-role-of-teachers-in-the-21st-century-education.
- Burnard, P., 1991. A method of analyzing interview transcripts in qualitative research.

 Nurse Education Today, 11, 461-466. doi:10.1016/0260-6917(91)90009-Y
- Cambridge University Reporter. (2003). Indicators of academic performance. Retrieved on August 8, 2017 from http://www.admin.cam.ac.uk/reporter/2002-3/weekly/5913/
- Kathy C., Kappan P.D., 2005 changing the Nature of Parental Involvement. Bloomington, 86(9): 645-646.
- Charoenruk, D., 2006. Communication Research Methodologies: Qualitative and Quantitative Methodology. Retrieved from:
- http://utcc2.utcc.ac.th/localuser/amsar/PDF/Documents49/quantitative_nd_qualitative_methodologies.pdf
- Chaturvedi, 2009. School Environment Achievement Motivation and Academic Achievement.Indian Journal of Social Science Researches 6(2),29-37, Selection Centre Central, Bhopal.
- Considine, G. and Zappala, G., 2002. Influence of social land economic disadvantage in the academic performance of school students in Australia. Journal of Sociology, 38, pp.129-148.
- Cohen, L. Manion, L. and Morris, K., 2007. Research methods in education (6th ed). London: Routledge.
- Creswell, J.W. 2003. Research Design Qualitative, Quantitative, and Mixed Method Approaches. Sage Publications, Thousand Oaks.
- Creswell, J.W. 2007. Qualitative inquiry and research design: Choosing among five approaches (2nd ed.).
- Crosnoe, R., Johnson, M. K. and Elder, G. H., 2004. School size and the interpersonal side of education: An examination of race/ethnicity and organizational context. *Social Science Quarterly*, 85(5), pp.1259-1274.
- Curriculum Research and Development Division of Ghana Education Service (2008). Visual Arts Teaching Syllabus. Accra.

- Daniels, T.O., 2005. Factors That Affect the Teaching and Learning Process in Primary Schools in Bundalang'i Division of Busia District. Unpublished M Ed Research Thesis, Nairobi, CUEA.
- Denscombe, M.2000. Forskringshand broken. Lund. Studentslitteratur.
- Devadoss, S., and Foltz, J., 1996. "Evaluation of Factors Influencing Students Attendance and Performance". American Journal of Agricultural Economics, 78(3), 499 507.
- Dorleku, A., 2013. Teaching and learning in border towns: A study of four junior high schools along the Ghana-Togo Border (Master's thesis). Department of General Art Studies, Kwame Nkrumah University of Science and Technology, Kumasi, Ghana.
- Durden, G. C. and Ellis, L. V., 1995. The effect of attendance on student learning in Principles of Economics. American Economic Review, 85, pp.343-346.
- Eamon, M. (2005). Social-demographic, school neighbourhood and parenting influences on academic achievement of Latino young adolescents. Journal of youth and Adolescence.
- Esterby-Smith, M., R. Thorpe and Lowe, A., 2002. Management Research: An Introduction (2nd Edition), London, Sage Publication.
- Ehren, M. C. M., and Visscher, A. J., 2008. The relationship between school inspections, school characteristics and school improvement. British Journal of Educational Studies, 56(2). 205-227.
- Ehiametor, E. T., 1990. Business and Economic Seduction, Principle and Practice:

 Lagos Evan BrasLtd
- Emerson, C. and Goddard, I., 1993. Managing staff in school. Oxford: Heinemann Educational
- Emory, C. W., and Cooper, D. R., 1991. Business research methods, (4th Ed.). Boston, MA: Irvin.
- Evans-Solomon, F. and Opoku-Asare, N.A. 2011. Girls' Motivation, Participation And Preference For Visual Arts Subjects In Four Senior High Schools In Central Region, Ghana Journal of Science and Technology, Vol. 31, No. 3 (2011), pp & 118-128

- Evans-Solomon, F., 2004. Girl-child education in the visual arts: Opportunities and challenges, (Unpublished Master's Thesis). University of Science and Technology, Kumasi.
- Farrant, J 1992, Principles and Practices of Education. Singapore: Longman Publishers PTE Ltd.
- Farrant, J. S., 1980. Principles and practice of education, (New ed.). Malaysia: Longman
- Felder, R. M., 1993. "Reaching the Second Tier: Learning and Teaching Styles in College Science Education". Journal of College Science Teaching, 23(5), 286 290.
- Foster, P., 1967. Education and Social Change in Ghana. Routledge and Kegan Paul. London
- Gagné, R.M & Merrill, M.D. 1990. Integrative Goals for Instructional Design. In: Educational Technology Research and Development, 38 (1), 23-30.
- Geshel Megan E., 2017. International Teachers`Perception of Leadership Practice For Effective Teacher Evaluation: A Qualitative Case Study.Ed.D, Dissertaions.95 http://commons.cu/Portland.ed/edudissertations/95.
- Ghauri P, and Gronhaug K., 2005. Research methods in business studies A practical guide. 3rd edition. Prentice Hall.
- Graetz, B., 1995. Socioeconomic Status in Education Research and Policy. In Ainley, J, Graetz, B., Long, M. and Batten, M. (Eds). Social economic Status and School Education. Canberra: DEET/ACER.
- Greene, B. A., and Miller, R. B., 1996. Influences on achievement: Goals, perceived ability, cognitive engagement. Contemporary Educational Psychology, 21, pp.181–192.
- Greene, B. A., Miller, R. B., Crowson, M., Duke, B. and Akey, K., 2004. Predicting high school students' cognitive engagement and achievement: Contributions of classroom perceptions and motivation. Contemporary Educational Psychology, 29, pp.499–517.
- Haist, S.A., Wilson, J.F., Elam, C.L., Blue, A.V., and Fosson, S.E. (2000). "The Effect of Gender and Age on Medical School Performance: An Important Interaction. Advances in Health sciences Education, 5(3), 197 205.

- Halcomb, E. J., Gholizadeh, L., DiGiacomo, M., Phillips, J., & Davidson, P. M. (2007). Literature review: Considerations in undertaking focus group research with culturally and linguistically diverse groups. Journal of Clinical Nursing, 16, 1000–1011
- Halsay, P.A., 2004. Nurturing Parental Involvement: Two middle level teachers share their secrets. Clearing House, 77(4); 135-157.
- Harb, N., and El-Shaarawi, A., 2006. "Factors Affecting Student Performance". Munich Personal RePEc Archive Paper No. 13621. Accessed on January 02, 2018 from http://mpra.ub.uni-muenchen.de/13621/.
- Hammond L.D., Austin, K., Orcutt, S. and Rosso J., 2001. How people learn: Introduction to learning theories. [online] Retrieved 17th Jul, 2017, from http://www.stanford.edu/class/ed269/hplintrochapter.pdf.
- Hollensen, S., 2007. Global Marketing: Fourth Edition, (4 ed.), Pearson Education Limited, Essex, England, P.62.
- Holme, I.M. and Solvang, B.K. 1997. Forskningsmetodik: Om kvalitativa och kvantitativa metoder, Lund: Studentlitteratur.
- Hoyle, E., 1986. Policies of School Management, Suffolk. The press ltd.
- Huei-Wen, L. and Huei-Fu, L., 2017. The evaluation of Event Sport Tourism on Regional Economic Development. International Journal of Science, Behavioral, Educational, Economic, Business and Industrial Engineering, Vol:11, No. 1.
- Jeynes, W., 2002. Examining the effects of parental absence on the academic achievement of adolescents. Journal of Family and Economic issues.
- Jolivette, K., Scott, T. M., Nelson, C. M., 2000. The link between functional behavioural assessments and behavioural intervention plans (Eric EC Digest No. E592). Arlington, VA: The Eric Clearinghouse on Disabilities and Gifted Education.
- Katana, 2007. Factors affecting performance in KCPE in Magarini division of Malindi District, unpublished Med project. UoN.
- Karen, B., 2011. In-depth Interviewing: The process, skill and ethics of interviews in peace research.
- Keung, S.T., 2014. Examining Academic Performance of Polynesian Student-Athletes Using the Theory of Planned Behavior. Master of Science Thesis.

- Kitavi, M. J., 2005. An investigation of the factors influencing performance in the Kenya certificate of primary education. Unpublished MEd thesis, University of Nairobi.
- Koontz, H and Weihrich, H., 1988. Management.McGraw-Hill, INC. New York
- Kyoshaba, M., 2009. Factors Affecting Academic Performance of Undergraduate Students at Uganda Christian University. Makerere University M. A. thesis.
- Kwesiga, C. J., 2002. Women's access to higher education in Africa: Uganda's experience.Kampala: Fountain publishers Ltd.
- Lloyd,C. and Ghuman, S., 2010. Teacher absence as a factor in gender inequalities in access to primary schooling in rural Pakistan Comparative Education Review, 54(4),539-554.
- Malka, A., and Covington, M. V., 2005. Perceiving school performance as instrumental to future attainment: Effects on graded performance. Contemporary Educational Psychology, 30(1), 60–80.
- Mbelle, A. and Katabaro, J., 2003. (University of Dar es Salaam) School Enrolment, Performance and Access to Education in Tanzania. Research On Poverty Alleviation Mkuki Mkuki Na Nyota Publishers Tanzania
- Miller-Grandvaux, Y. and Yoder, K., 2002. A literature review of community schools in Africa. USAID, Bureau for Africa. Washington DC. 113pp.Educational Psychology., 29 (4) (2004), pp. 462-482
- Miller, R.B., Greene, B. A., Montalvo, G.P., Ravindran, B., Nichols J. D., 1996. Engagement in Academic Work: The Role of Learning Goals, Future Consequences, Pleasing Others, and Perceived Ability Contemp Educ Psychol. 21(4):388-422.
- Mlambo, V., 2011. An analysis of some factors affecting student academic performance in an introductory biochemistry course in the University of the West Indies. *Caribbean Teaching scholar*, 1(2), 79-92.
- MolokoMphale, L. and Mhlauli, M.B., 2014. An Investigation on Students Academic Performance for Junior Secondary Schools in Botswana. European Journal of Educational Research. Vol. 3, No. 3, pp.111-127

- Mordedzi, B.D.P. 1999. Social reproduction, transformation, and change: Implications for business education. Mate Mase Journal of the University College of Education of Winneba, 1, 37-50.
- Morgan, D. L., 1996. "Focus Groups." Annual Review of Sociology. 22, pp.129-152.
- Morrow, W. (2007). Learning to teach in South Africa. Cape Town, South Africa: HSRC Press.
- Moore,D.1998. Improve your school atmosphere. School planning and Management journal,37(10), 18
- Mortimore, P.1998. The Roard to Improvement: Reflection in School Effectiveness, Lisse: Swets and Zeitlenes.
- Malaysia'. Canadian Research and Development Center of Sciences and Cultures: Vol.3 No.4.p1
- Mushtaq, I. and Khan, S.N., 2012. Factors Affecting Students' Academic Performance Global Journal of Management and Business Research Volume 12 Issue 9 Version 1.0 June 2012 Online ISSN: 2249-4588 and Print ISSN: 0975-5853
- Newman-Ford, L., Lloyd, S., and Thomas, S., 2009. "An Investigation in the Effects of Gender. Prior Academic Achievement, Place of Residence, Age and Attendance on First-year Undergraduate Attainment". Journal of Applied Research in Higher Education, (1(1), 13-28.
- Neuman, W. L., 1997. Social research methods-qualitative and quantitative approaches. Needham Heights: Allyn & Bacon.
- Norhidayah, A., Kamaruzaman, J., Syukriah, A. Najah, M. and Andin, S.A.S., 2009. 'The Factors Influencing Students' Performance at Universiti Teknologi MARA Kedah.
- OFSTED. 1994. Secondary OFSTED Working Papers for Inspection. London: HMSO.
- Ormrod, J. E., 2008. Educational psychology: Developing learners (6th ed.). Upper Saddle River, NJ: Pearson Education.
- Opoku-Asare, N. A., 2008. Promoting quality visual arts education through teacher effort and education: Continuing professional education seminar paper: Department of Art Education, KNUST, May 2008.

- Orenge, M. J., 2007. Analysis of Factors which Contribute to Poor Students

 Performance in KCSE Examinations In Nyacheki Division ,Gucha District

 Kenya, Unpublished Med project, University of Nairobi.
- Osei-Mensah, F., 2012. Factors That Influence The Performance In General Knowledge In Art Of Senior High School Students In Abura Asebu Kwamankese District In The Central Region. KNUST Master of Arts in Art Education Thesis
- Otoo, D., 2007. Comparative study of academic performance of public and private JSS graduate: A case study of four selected senior secondary schools in the Kumasi Metropolis (Master's thesis). Centre for Educational Policy Studies, University of Education, Winneba, Ghana.
- Owusu-Afriyie, C., 2009. The school administrator as an instructional leader of the visual arts programme: A case study of Ejisu-Juaben district of Ashanti Region (Master's thesis). Department of General Art Studies, Kwame Nkrumah University of Science and Technology, Kumasi, Ghana.
- Pedrosa, R. H. L., Norberto J.W., Rafael P., Maia D., Andrade, C.Y. & Carvalho, B. S., 2006. Educational and Social economic background of graduates and academic performance. Brazilian Research University. Presented at the IMHE/OECD General Conference, Paris, September, 2006
- Park, K.H., and Kerr, P. M., 1990. "Determinants of Academic Performance: A Multinomial Logit Approach". Journal of Economic Education, 21(2), 101-111.
- Pashler, H., McDaniel, M., Rohrer, D., and Bjork, R., 2008. "Learning Styles: Concepts and Evidence". Psychological Science in the Public Interest, 9(3), 106 119.
- Pintrich, P. R., 1999. The role of motivation in promoting and sustaining self-regulated learning. International Journal of Educational Research, 31, pp.459–470.
- Pintrich, P. R. and Schrauben, B., 1992. Students' motivational beliefs and their cognitive engagement in classroom academic tasks. In D. Schunk and J. Meece (Eds.), Student perception in the classroom, pp.149–179. Hillsdale, NJ: Earlbaum.

- Pintrich, P. R. and Schunk, D. H., 2002. The role of goals and goal orientation. In P. R. Pintrich and D. H. Schunk (Eds.), Motivation in education: Theory, research and application (2nd ed., pp.190–242). Englewood Cliffs, NJ: Simon and Schuster.
- Reddy, P and Talcott, J. 2006. Predicting university success in psychology: Are subject-specific skills important? Retrieved on July 4, 2017. from http://www.aston.ac.uk/downloads/ihs/peelea/huw2006p.pdf.
- Reid, J. M. 1995. "Learning Styles in the ESL/EEL Classroom". Heinle & Heinle Publishers: Boston.
- Remenyi, D., Williams, B., Money, A. & Swartz, E., 2003 Doing Research in Business and Management: An Introduction to Process and Method , London, SAGE Publications.
- Rihani, M. A., 2006. Keeping the promise: Five benefits of girls' secondary education. Washington, DC: Academy for Educational Development.
- Roberts, G. A., 2007. The effect of extracurricular activity participation in the relationship between parent involvement and academic performance in a sample of third grade children.
- Robson, C., 2002. Real World Research: A Resource for Social Scientists and Practitioner-Researchers. Oxford: Blackwell
- Rodgers, J. R., and Rodgers, J. L., 2003." An Investigation into the Academic Effectiveness of Class Attendance in an Intermediate Microeconomic Theory Class". Faculty of Commerce Papers, University of Wollongong, N.S.W., 2522 Australia. Access on January 02, 2018 from http://ro.uow.edu.au/commpapers/174.
- Romer, D., 1993. "Do Students Go to Class? Should they?". Journal of Economic Perspectives 7(3), 167 174
- Saxton, J., 2000. Investment in education: Private and public returns. Retrieved from http://www.house.gov/jec/educ.pdf.
- Saunders, M., Lewis, P., and Thornhill, A., 2007. Research methods for business students, (5th ed.). England: Pearson Educational Limited.
- Schmidt, R.M., 1983. Who maximizes what? A study in student time allocation. The American Economic Review, 73(2), pp.23-28.

- Schultz, T.P., 2002, "Wage gains associated with height as a form of health human capital", American Economic Review 92 (2), 349-353.
- Schunk, D. H., 1991. Self-efficacy and academic motivation. Educational Psycholigist, 26, pp.207–231.
- Sentamu, N.P., 2003. School's influence of learning: A case of upper primary schools in Kampala and Wakiso Districts. Uganda Education Journal, 4.
- Shoqirat, N., 2009. The role of Jordanian hospital in promoting patients' health.

 PhD theses. Queen Margaret University.
- Shultz, T.P. "Why Governments should Invest More to Educate Girls", World Development, Vol. 30 No. 2 (February, 2002) pp. 207-226
- Siaw, A. O. 2009. A comparative study of teaching and learning processes of the visual arts in selected senior high schools in urban and rural settings in Ashanti Region, Ghana (Master's thesis). Department of General Art Studies, Kwame Nkrumah University of Science and Technology, Kumasi, Ghana.
- Snow, C. E., Burns, S., & Griffin, P. (Eds.), 1998. Preventing reading difficulties in young children. Washington, DC: National Academy Press.
- Solomon LJ, Rothblum ED., 1984. Academic procrastination: Frequency and cognitive-behavioral correlates. J Couns Psychol 31(4): 503.
- Sparkes, J., 1999. Schools, Education and Social Exclusion, CASE Paper 29, Centre for Analysis of Social Exclusion, London School of Economics, London.
- Squires, G., 2002. Managing your learning. London, England: Routledge.
- Strauss, V., 2013. School principals and the rhetoric of 'instructional leadership'.

 Retrieved 27 August 2017 from: http://www.washingtonpost.com/blogs/answersheet/wp/2017/04/18
- Taal, A. H. S., 1996. Teacher education and training in Sub-Saharan Africa: Teacher education in Africa, past, present and future.Dakar: UNESCO.
- The President's Committee on Review of Education Reforms in Ghana. (2002). Meeting the challenges of education in the 21st century. Accra, Ghana: Adwinsa Publications.
- Tsinidou, M., Gerogiannis, V. and Fitsilis, P., 2010. Evaluation of the factors that determine

- quality in higher education: an empirical study. *Quality Assurance in Education*, 18(3), pp.227-244.
- UNESCO. 2001. Cultural heritage, creativity and education for all in Africa; for education in the arts and creativity in primary and secondary schools. Regional Conference on Arts Education, Port Elizabeth, South Africa.
- UNESCO. 2007. EFA. Global Monitoring Report: The Role of the Organization and Social Context of Schools. http://portal.org/education.
- UNESCO. 2008. Gender and Education for All: the Leap to Equality, Summary Report. UNESCO, Paris Waudo, J. and Ouya, E. (2010), Total Quality Management in Education, Rinny Educational and technical publishing services, Nairobi.
- Vrugt, A., Langereis, M. and Hoogstraten, J., 1997. Academic self-efficacy and malleability of relevant capabilities as predictors of exam performance. Journal of Experimental Education, 66, pp.61–74.
- Vrugt, A., Oort, F. J. and Zeeberg, C., 2002. Goal orientations, perceived self-efficacy and study results amongstbeginners and advanced students. British Journal of Educational Psychology, 72, pp.385–397.
- Wallace Foundation, 2011. The school principal as leader: Guiding schools to better teaching and learning. Retrieved from http://www.wallacefoundation.org/knowledge-center/school leadership/effective-principal-leadership/Documents/The-School -Principal-as-Leader-Guiding-Schools-to-Better-Teaching-and -Learning.pdf
- Walberg, H. J., 1981. A psychological theory of educational productivity. In F. H. Farley and N. U. Gordon (Eds.), *Psychology and education*. Berkeley, CA: McCutchan.
- Wenglinsky, H., 2001. Teacher Classroom Practices and Student Performance: How Schools Can Make a Difference Statistics and Research Division Princeton, NJ 08541.RESEARCH REPORT September 2001 RR-01-19.
- Wiid, J. and Diggines, C. 2009. Marketing research. Juta, Cape Town.
- Witziers, B., Bosker, R. J., and Krüger, M. L., 2003. Educational leadership and student achievement: The elusive search for an association. Educational Administration Quarterly, 39(3), 398-425. DOI: 10.1177/0013161X03253411

- Witkin, H.A., Oltman, P., Goodenough, D., Friedman, F., Owen, D. and Raskin, E. (1977b) Role of Field Dependent and Field Independent Cognitive Styles in Academic Evolution: a longitudinal study, Journal of Educational Psychology, 69, pp. 197-211.
- Witkin, H.A. and Goodenough, D.R., 1981 Cognitive Styles: essence and origins, field dependence and field independence, Psychological Issues, 14, Whole issue No. 51.
- World Bank. 2009. Expanding Opportunities and Building competencies for young people. Anew Agenda for secondary Education, Washington DC: World Bank.
- Woodfield, R., and Earl-Novell, S., 2006. "An Assessment of the Extent to which Subject Variation in Relation to the Award of First Class Degree between the Arts and Sciences can explain the 'gender gap". British Journal of Sociology of Education, 27(3), 355 372.
- Woodside, A. G., 2010. Case Study Research methods for theory building.: Theory, Methods, Practice. Bingley, Emeralds. Journal of Business and Industrial Marketing 18 (6/7), 493-508, 2003. 526, 2003.
- Yeboah, K., 2014. The push and pull factors motivating senior high school teachers in rural areas: A case study in AfigyaKwabre West District of Ashanti Region, Ghana (Master's thesis). Department of General Art Studies, Kwame Nkrumah University of Science and Technology, Kumasi, Ghana.
- Yin, R. K., 1994. CASE STUDY RESEARCH Design and Methods Second Edition.

 Applied Social Research Methods Series Volumes SAGE Publications

 International Educational and Professional Publisher Thousand Oaks London

 New Deihl Yin, Robert K., (2003b).3rd edition.Case Study Research: Design

 and Methods, Sage, Thousand Oaks, CA,
- Yusuf, A. F., 2012. Influence of principals' leadership styles on students' academic achievement in secondary schools. Journal of Innovative Research in Management and Humanities, 3(1), 113-121
- Zapalska, A.and Dabb,H.2002. Learning Style. International Business Teaching in Eastern and Central European Countries.13(3/4).77-97.

- Zikmund, WG (2000). Exploring marketing research, 7th edn, Dryden Press, Forth Worth.
- Zikmund, G. and Babin, B. J. 2010. Exploring marketing research. 10th ed. China: South Western.
- Zimmerman, B. J., and Kitsantas, A., 2005. Homework practices and academic achievement: The mediating role of self-efficacy and perceived responsibility beliefs. Contemporary Educational Psychology, 30(4), 397-417.



APPENDIX 1

DISCUSSIONS THAT EMERGED WITH THE STUDENTS DURING FOCUS
GROUP DISCSSION. POSING THE FOLLOWING QUESTIONS TRIGGERED THE
DISCUSSIONS THAT ENSUED BETWEEN THE INVESTIGATOR AND
PARTICIPANTS

(1). What do you understand by the term academic performance?

PART B: Factors that contribute to Visual Arts students' low academic performance.

- (2). What do you understand by the term low academic performance?
- (3). Do you agree to the fact that certain factors contribute to low academic performance?
- B2: Parents or guardians factors
- (4). Were your parents involved in your studies?
- (5). Through what means did your parents became involved in your learning?(This question was posed to the participants who answered "yes" in question 4 above.)
- (6). Why do you think your parents weren't involve in your studies?(*This question was posed to the participants who answered "no" in question 4 above.*)
- B3: Type or nature of School

- (7). In what way or ways do you think the prevailing conditions and facilities in Dormaa Senior High School Visual Arts Department contributed to your low academic performance?
- (8). In what way or ways do you think other schools are ahead of Dormaa Senior High School in Visual Arts and thus contribute to its Visual Arts students' low academic performance?

B4: Class attendance and academic performance

- (9). What way or ways do you agree to the fact that truancy contributes to students' low academic performance?
- (10). What are the major reasons that normally account for truancy on the part of Visual Arts students in Dormaa Senior High School?

B5: Entry qualifications and prerequisites

- (11). Comparing your entry qualifications to your counter parts in the other elite schools, will you agree that you were admitted based on superior qualification than them?
- (12). So what accounted for your low academic performance in the Visual Arts Programme you were offered by Dormaa Senior High School?(*This question was posed to participants who responded "yes" to question 11 above*).
- (13). Though you entered the Dormaa Senior High School with such a qualification, what actually contributed in your low performance for the Visual Arts Programme?

(14). So what accounted for your low academic performance in the Visual Arts Programme you were offered by Dormaa Senior High School?(*This question was posed to participants who responded "no" to question 11 above*).

PART C: Strategies that can be put in place to improve students' academic performance

- (15). Can you recommend to the leadership of this school, strategies that they can put in place to help improve the academic performance of visual arts students?
- 16) What role should the students play to enhance academic performance of visual Arts students in Dormaa Senior High School?
- 17) What role should the Dormaa Ahenkro community play to enhance academic performance of visual Arts students in Dormaa Senior High Sshool?
- 18) What role should the old students play to enhance academic performance of Visual Arts students in Dormaa Senior High School?

APPENDIX 2

INTERVIEW QUESTIONS FOR THE FOUR TEACHERS

The interview focused on School Leadership and teachers' contribution to low students' academic performance in the Visual Arts Department

PART A: FACTORS THAT CONTRIBUTE TO LOW ACADEMIC PERFORMANCE

- (1). Do you agree to the fact that the school leadership have an impact on the students' academic performance?
- (2). In specifics, through what means can you say the school leadership have impact on the students' academic performance?
- (3). What comprise the school leadership involvement in instructional leadership?
- (4). An effective instructional leadership is able to build the culture of learning, provide resources needed for quality teaching, and structure the classroom environment to stimulate teachers and learners motivation. Do you see these activities evidenced in this school?
- (5). Do you agree to the fact that this school for the past three years has registered low students' academic performance in your department?
- (6). What are the factors that will account for the low students' academic performance in this school?
- (7). Do you agree to the fact that teachers in general contribute greatly in students' academic performance (either low or high)?

- (8). Through what means can you say teachers in this school have been contributing to low students' academic performance attained by Visual Arts students?
- (9). It is a general feeling that students who fail the examinations are taught by ineffective teachers; on the other hand those who excel are taught by the very effective teachers. In what way or ways will you explain that teacher's effectiveness has an influence on the students' academic attainment?

PART B: Strategies that can be put in place to improve students' academic performance

- (1). Can you recommend to the leadership of this school, strategies that they can put in place help improve the academic performance of visual arts students?
- (2). Can you recommend to your fellow teachers Factors that can contribute to their effective functions?